Burnout: An exploration of burnout experiences by providers of homeless services in Ireland

Marta Franchy Garcia

Master of Arts in Human Resource Management

National College of Ireland

Submitted to the National College of Ireland, May 2021

Abstract

Burnout has become one of the main issues experienced in the work development nowadays, which has gathered a large body of research in the last decade. The gross of the research carried out in this field in the Republic of Ireland focalises on three main groups: the healthcare sector, social work with a predominance of the child protection division and the educational field. Leaving the provision of homeless services understudied

Understanding Burnout in relation to the homeless service provision takes a principal stand in the Irish territory as the nation has been experiencing a homeless crisis for the past ten years. Positioning the providers of this type of human services under a large amount of pressure while working with a sector of the population characterised by their chaotic nature, resulting from the traumatic experiences suffered, such as alcoholism, drug addiction or sexual assault.

This research was developed due to the lack of research on burnout perceptions by providers of homeless services, and the aim to provide an understanding of burnout in this sector concerning the three dimensions of job burnout (emotional exhaustion, cynicism and inefficacy), the stress suffered, and the job demands and resources impacting its incidence.

For this matter, and to provide a comprehensive understanding of the burnout experiences of homeless service providers, a qualitative approach to research was carried out. Twelve employees of the organisation framing this study, Focus Ireland, provided their accounts regarding the research object through semi-structured interviews.

The findings of this research elucidate that in the provision of homeless services, only two of the burnout's dimension are experienced, emotional exhaustion and cynicism, while there is no evidence of the presence of inefficacy within this sector. Furthermore, the study carried out also finds evidence that supports the conception of the homeless sector as a highly stressful field of employment; and the existence of job demands and resources attached to the development of their work that might be inciting the prevalence of burnout in this field.

Submission of Thesis and Dissertation

National College of Ireland

Research Students Declaration Form

(Thesis/Author Declaration Form)

Name: Marta Franchy Garcia

Student Number: x19126158

Degree for which thesis is submitted: MA in Human Resource Management

Title of Thesis: Burnout: An exploration of burnout experiences by providers of homeless services in Ireland

Date: 5th May 2021

Material submitted for award

(a) I declare that the work has been composed by myself. \checkmark

(b) I declare that all verbatim extracts contained in the thesis have been distinguished by quotation marks and the sources of information specifically acknowledged. \checkmark

(c) My thesis will be included in electronic format in the College Institutional Repository TRAP (thesis reports and projects). \checkmark

(d) **Either** *I declare that no material contained in the thesis has been used in any other submission for an academic award. **Or** *I declare that the following material contained in the thesis formed part of a submission for the award of Master of Arts in Human Resource Management. \checkmark

Signature of research student:

Date: 5th May 2021

Acknowledgements

I would like to acknowledge and extend my gratefulness to the following people who have helped me complete my studies and this dissertation.

First and foremost, my supervisor Desmond Gibney, who supported and guided me in developing and completing this dissertation.

All the Focus Ireland participants in this study without whom this research could not have been completed.

My college peers Alex, Fauve, Marian and Marta, who started this journey with me and have been a great source of support and encouragement along the way.

My partner David, who has held the fort for me at home so I can dedicate myself to my studies and this research.

My dearest friends Rachel and Patri for giving me the reassurance and moral support needed throughout this process.

And last but not least, I would like to thank my parents Fabiola and Carlos, for always supporting my ventures and believing that I can achieve everything I put my mind and heart into.

Table of Content

| Abstract 2 |
|--|
| Acknowledgements |
| List of Acronyms |
| Chapter 1. Introduction9 |
| 1.1 Overview of the Research |
| 1.2 Dissertation Subject |
| 1.3 The Irish Homeless Context |
| 1.3.1 Defining Homelessness |
| 1.3.2 Homeless Statistics |
| 1.4 Organisational Context |
| 1.5 Purpose and Relevance of the Research13 |
| 1.6 Configuration of the Reminder Dissertation13 |
| Chapter 2. Literature Review15 |
| 2. Introduction15 |
| 2.1 Conceptualisation of Burnout15 |
| 2.1.1 Emotional Exhaustion |
| 2.1.2 Cynicism |
| 2.1.3 Inefficacy |
| 2.2 Stress |
| 2.3 Antecedents of Burnout |
| 2.4 Job Demands-Resources Model20 |
| 2.4.1 Job Demands |
| 2.4.2 Job Resources |
| 2.5 Impact of Burnout |
| 2.5.1 Individual Effects |
| 2.5.2 Organisational Effects |
| 2.6 Provision of human services |
| 2.6.1 Homeless Services |
| 2.7 Summary |
| 2.8 Research Aim and Questions29 |
| Chapter 3. Methodology |
| 3. Introduction |
| 3.1 Research Philosophy |

| 3.2 Research Approach | 32 |
|---|----|
| 3.3 Research Design | 33 |
| 3.4 Research Tool | 34 |
| 3.5 Sample | 35 |
| 3.6 Pilot Study | 37 |
| 3.7 Analysis of Data | 38 |
| 3.8 Ethical Considerations | 40 |
| 3.8.1 Pre- and During the interview | 41 |
| 3.8.2 Data and Storage | 41 |
| 3.8.3 Partakers | 41 |
| Chapter 4. Findings | 43 |
| 4. Introduction | 43 |
| 4.1 Theme 1. Identification of Burnout | 43 |
| 4.2 Theme 2. Emotional Exhaustion | 44 |
| 4.2.1 Sub Theme. Physical and Psychological Fatigue | 44 |
| 4.2.2 Sub Theme. Energy Replenishment | 44 |
| 4.3 Theme 3. Cynicism | 45 |
| 4.3.1 Sub Theme. Detachment from Clients | 45 |
| 4.3.2 Sub Theme. Organisational Engagement | 46 |
| 4.3.3 Sub Theme. Sick Days and Extra Breaks | 46 |
| 4.4 Theme 5. Stress | 48 |
| 4.4.1 Sub Theme. Workload | 49 |
| 4.4.2 Sub Theme. Role Ambiguity and Role Conflict | 50 |
| 4.4.3 Sub Theme. Emotional Demands | 51 |
| 4.4.4 Sub Theme. Working Hours | 51 |
| 4.4.5 Sub Theme. Peer Support | 52 |
| 4.4.6 Sub Theme. Organisational Support | 52 |
| 4.4.7 Sub Theme. Supervision | 53 |
| 4.4.8 Sub Theme. Pay | 54 |
| 4.5 Limitations of the Study | 55 |
| Chapter 5. Discussion | 56 |
| 5.1 Burnout Dimensions | 56 |
| 5.1.1 Emotional Exhaustion | 56 |
| 5.1.2 Cynicism | 57 |
| 5.1.3 Inefficacy | 58 |
| 5.2 Stress | 58 |

| 5.3 Job Demands and Job Resources | 59 |
|---|-----|
| 5.3.1 Job Demands | 59 |
| 5.3.2 Job Resources | 61 |
| Chapter 6. Conclusion | 64 |
| 5.4 Suggestions for Future Research | 65 |
| Chapter 7. CIPD Requirements | 67 |
| 7.1 Recommendations | 67 |
| 7.2 Statement of Personal Development | 68 |
| References | 70 |
| Appendices | 88 |
| Appendix A- Interview Questions and Questions Rationale | 88 |
| Appendix B - Participation Request Email | 101 |
| Appendix C - Participant Information Sheet | 102 |
| Appendix D - Consent Form | 104 |

List of Acronyms

| AI | Advice and Information |
|---------|---|
| СМ | Case Management |
| DRHE | Dublin Region Homeless Executive |
| FEANTSA | European Federation of National Organisations Working with the Homeless |
| JDR | Job Demands-Resources |
| MBI | Maslach's Burnout Inventory |
| WHO | World Health Organisation |

Chapter 1. Introduction

1.1 Overview of the Research

The purpose of this dissertation is to explore the perceived job burnout level and its causes within providers of homeless services in the Republic of Ireland.

The provision of homeless services is a challenging field due to the particular vulnerability of this group. Homeless individuals present high levels of stress and generally present a lack of social skills, financial means and social support, which adds an extra degree of difficulty in working with this group that no other social field experience with such intensity (Kidd *et al*, 2007). Throughout their lives, homeless individuals are often exposed to a wide range of traumatic circumstances such as domestic violence, addiction, neglect, psychological and sexual abuse, among many others. The nature of these experiences creates sentiments of helplessness and fear that overwhelms the individual and diminish their capacity to cope with their reality (Waegemakers Schiff and Lane, 2019). The circumstances described thereto highlight this group need for intensive support to accomplish different ordinary tasks, which increments the amount of pressure on the professionals involved with this group, making them more susceptible to developing stress-related conditions such as burnout (Mette *et al*, 2020).

1.2 Dissertation Subject

The central concept object of this dissertation is the burnout syndrome or burnout.

Burnout, understood as a response to a chronic exposition to work-related stressors, is a topic that has been gaining international attention due to the implications that it holds for organisational and personal wellbeing (Maslach, Schaufeli and Leiter, 2001). Burnout has been found to be a source of psychological and physical ailments for the individual and the cause for unfavourable work-related conditions such as absenteeism, low quality of work, and intention to leave, among many others (Suñer-Soler *et al*, 2014). The integral effects of burnout on an individual's life have made this phenomenon an essential subject of study in recent years. Since its first conceptualisation, burnout syndrome has been positively linked to professional development in human services due to the many challenges faced by those employed in this field, such as high caseloads, lack of role clarity, low remuneration, and limited prospects of upward promotions (Thomas, Kohli and Choi, 2014). Research carried out in the burnout field has highlighted the effects of such demanding environments on homeless service providers and have even estimated that in this field, the service providers' lifespan

is shorter than recognised highly stressed professionals, such as doctors, nurses or educators (Curtis, Moriarty and Netten, 2010).

Through this dissertation's development, an extensive examination and formulation of burnout will be carried out, formulating the different approaches to the concept developed through the years and the conditions that influence its manifestation and its consequences for both the organisation and the person. This dissertation will pay special attention to the experiences regarding burnout of the service providers in this sector.

1.3 The Irish Homeless Context

To completely understand the challenges experienced by homeless services providers and the particular characteristics that this group present, an analysis of the context they are immersed in is required. Therefore, an examination of the concept and state of the homelessness phenomenon in Ireland will be carried out.

1.3.1 Defining Homelessness

Homelessness is a broad concept that has been subjected to numerous designations and classifications throughout history without providing a universal description to identify and categorize this phenomenon. The definition of this subgroup of society continues to vary in characterization nowadays.

In the Republic of Ireland, the legal definition of homelessness is stated in the *Housing Act 1988*. For the purpose of the Act, a person is regarded as homeless if:

"(a) there is no accommodation available which, in the opinion of the authority, he, together with any other person who normally resides with him or who might reasonably be expected to reside with him, can reasonably occupy or remain in occupation of,

or (b) he is living in a hospital, county home, night shelter or other such institution, and is so living because he has no accommodation of the kind referred to in paragraph (a), and he is, in the opinion of the authority, unable to provide accommodation from his own resources."

The legal definition of homeless, has been the classification accepted and adopted by the local authorities responsible of the provision of homeless services in Ireland such as the Dublin Region Homeless Executive (2021), henceforward DRHE.

According to research carried out by the European Federation of National Organisations Working with the Homeless, FEANTSA (2014), the Republic of Ireland does not recognise as homeless, individuals

institutionalised, people living in unconventional dwellings due to the lack of conventional housing, and people residing temporarily with family or friends. This exclusion provides for a data recording problem for local authorities as people and families residing in unconventional dwellings or with families and friends are also recipients of homeless support through prevention services and, thus are part of the homeless services' customer base. The lack of inclusion of these three subgroups of individuals described thereto illustrates that the national agencies' statistics in terms of individuals recognized as homeless might be higher than stated.

Correspondingly, FEANTSA, developed a European Typology of Homelessness and housing exclusion (ETHOS) to provide a common designation for homelessness for international interactions, such as debates, measurement and policy development. ETHOS provided for an in-depth characterisation of homelessness resulting of the different designations and experiences of this phenomenon through Europe. The definition of homeless elucidated within the ETHOS framework, developed by FEANTSA (2005), recognises as homeless:

- People sleeping rough
- People residing in temporary accommodation such as shelters or refuges
- People living in unstable dwellings, threatened by eviction or domestic violence.
- People living in inadequate accommodation such as overcrowding or mobile homes.

This broader and more inclusive definition of homelessness has been adopted by voluntary organisations providing homeless support in the national territory, such as Focus Ireland (2021) and Simon Communities of Ireland (2021).

1.3.2 Homeless Statistics

The homeless state in the Irish territory has been regarded by the Government of Ireland (2018) as an ever-increasing crisis, placing this social problem as one of the main priorities for the current government.

In recent research, the Government of Ireland (2021), through the Department of Housing, Local Government and Heritage, reported 5,987 adults accessing homeless emergency services in Ireland during January 2021. Furthermore, this report also provides a recount of the families accessing emergency services nationwide, with 966 families, 1,430 adults, and 2,326 dependants. Moreover, the DRHE (2019) has also documented through research the existence of at least 128 rough sleepers in the area of Dublin, which are not counted in the national statistics as they do not avail of homeless supports provided by local authorities. It is also noteworthy to mention that the official count of rough

sleepers only occurs in the Dublin area, leaving the counties with an undetermined number of rough sleepers.

As aforementioned, the numbers stated thereto comprehend exclusively the number of people accessing homeless accommodation services, with no national record of how many homeless people, or at risk of homelessness, access non-residential services, such as drop-in facilities or prevention services. Therefore, the number of homeless service users is expected to be higher than the numbers stated above, and thus the demand for homeless related services.

The existence of homelessness in every county of the Republic of Ireland offers a diverse scenario for the provision of homeless services. The dissemination of the homeless population offers the possibility to analyse the different circumstances providers of homeless services face in delivering their practice, and thus their experiences of burnout in different areas of the Irish territory. Therefore, the research carried out will not focalise in a concrete area of the national territory, allowing for the participation of homeless service's providers that operate nationwide

1.4 Organisational Context

The delivery of homeless services in the national territory is mainly provided by non-profit organisations and charities. These organisations offer different levels of interventions for people at risk or experiencing homelessness, such as residential facilities, drop-in services, food services, prevention and housing support, among many others. Often these organisations are provided with funding from local authorities, but in some instances, they operate with the sole fund of donations. Ireland counts with several non-profit organisations focused on the homeless issue, such as Crosscare, Peter McVerry Trust and Depaul Ireland, but the organisation chosen for this research is Focus Ireland.

Focus Ireland (2019) is one of Ireland's largest homeless organisations, which counts with about 400 employees across the territory and provides services in Carlow, Limerick, Clare, Monaghan, Cork, Sligo, Dublin, Tipperary, Kildare Waterford, Kilkenny and Wexford. The large scope of the organisation provides for a diverse scenario to explore the different experiences of burnout by employees under the same organisation across the Irish territory.

The organisation, Focus Ireland (2021), offers a diverse range of services for people at risk or experiencing homelessness, with prevention services, residential services, day services, advocacy and communication, and housing development services. The diverse landscape of services is one of the

critical components for choosing this organisation, as it will allow us to identify if the patterns of burnout presented in the organisation differ depending on the type of service.

Throughout 2019, Focus Ireland (2019) assisted a total of 14,200 individuals across the Irish territory, which exemplifies the organisation's enormous scope and the extensive work the employees carry out.

1.5 Purpose and Relevance of the Research

While there has been a limited amount of research in Ireland regarding burnout syndrome in the last decades. The studies that have been carried out predominantly relate to healthcare providers and social workers practising within the child and family agency (McFadden, 2020; McNicholas *et al*, 2020; McFadden, Mallet and Leiter, 2018; Sulaiman *et al*, 2017); Leaving other sectors, such as the homeless provision services, underexplored and in need of an in-depth analysis of the potential burnout being experienced. Therefore, the purpose of this research is to cover that gap and to provide a study of the homeless service provider's sentiments and experiences of burnout.

Similar research has been carried out in other countries to determine the existence of a link between the work carried out in homeless services and burnout. An example of this is the research carried out by Lemieux and Taylor in 2019. This study investigated the relationship between mental health strain and burnout on employees of third sector organisations, providing services for people at risk or experiencing homelessness in the United Kingdom. The findings of this research state the existence of a link between the level of stress experienced in this type of human services and the development of burnout (Lemieux and Taylor, 2019).

Considering the conclusions of this study, it could be of interest to analyse the context of the homeless services in Ireland, through the experiences of service providers to see if the same correlations regarding stress and burnout are reached.

1.6 Configuration of the Reminder Dissertation

The remainder of the dissertation is structured as follows.

Chapter 2 contains the literature review that focuses on an in-depth conceptualisation of burnout. An exhaustive analysis of the sources and consequences of burnout and the specific conditions that homeless services present will be carried out. The literature review will also encompass the relevant

research carried out regarding burnout in recent years and provide a foundation for addressing the research question identified. Furthermore, this chapter also provides the formulation of the research questions chosen to guide this study and the sub-questions and objectives attached to it.

Chapter 3 focuses on the methodology used and the research designed applied to the study carried out. This chapter provides the reader with an understanding of the methods of research available and the study design and tools applied to the research carried out. Furthermore, this chapter discusses the ethical concerns attached to this study.

Chapter 4 is constituted by the research findings originating from the interviews carried out, which have been divided into themes and subthemes, providing quotes from the participants to exemplify the findings encountered. Moreover, this chapter provides a discussion of the limitations identified in the research conducted.

Chapter 5 encompasses the discussion of the findings provided. This chapter provides a critical analysis of the findings generated and their relation to the academia provided in the literature review.

Chapter 6 provides the primary outcomes of this research and offers suggestions for future studies in the field.

Chapter 7 provides the requirements established by the Chartered Institute of Personnel Development (CIPD) for this dissertation.

Chapter 2. Literature Review

2. Introduction

The literature review aims to provide an understanding and gain knowledge in the research carried out on this specific topic, the burnout syndrome. An exploration of the concept of burnout with its dimensions, antecedents and impact will be carried out, focusing, for the purpose of this research, in the work related factors that produce this phenomenon. Furthermore, an examination of the burnout syndrome in homeless services will be revised to provide a conceptualisation of this specific working environment through previous research developed in this field.

2.1 Conceptualisation of Burnout

Burnout has been generally defined as a syndrome resulting from a continuous exposition to workrelated stress, which generally affects professions that work focuses on contact with other individuals. This syndrome was originally reported in care and human service professions whose aim is to provide for people in need and in which the interaction between professional and customer is paramount. This concept has become relevant for other professions outside of the social service providers throughout the years, and its research continues to grow globally. But to understand this concept in its totality, a review of its conceptualization will be carried out.

The concept of burnout first appeared, as referenced by Maslach *et al* (2001), in Graham Greene's novel "A burn-out case". In his novel, Greene (1961) narrates the story of an architect that has a psychological breakdown and decides to move to Congo, exemplifying through this fable how extreme fatigue can turn into irritation and despitefulness towards a person's job.

This concept will not find its course in scientific literature until the 1970s. Freudenberger (1974) first conceptualised this syndrome as the state of fatigue or frustration created by the work's demands, presented among professionals that experienced intense interaction with their clients (Freudenberger, 1974). Freudenberger's pioneering work opened the field for numerous studies on the subject in different fields in the subsequent years and established him as the concept's founding father.

This first conceptualisation was followed by numerous studies in the field that provided definitions of what burnout constitutes and the conditions involved in its manifestation. Maslach (1978) defined

burnout as a psychological syndrome developed as a response to chronic interpersonal stress, in which the service provider is unable to maintain the caring state brought initially to the job and loses all positive feelings for the clientele. Whereas Pines and Kafry (1978) defined burnout as a breaking point in which the work demands surpass one's endurance and capability to cope due to the overpowering work-related stress. Similarly, Edelwich and Brodsky (1980) asserted a definition of burnout as an evolving process in which a person acute a loss of idealism, energy and purpose, derived from their working conditions and the stress perceived. In which there are four identifiable stages: enthusiasm, stagnation, frustration and apathy. Correspondingly, Perlman and Hartman (1982) carried out a summary of the different theoretical constructions of burnout, creating a definition that included work-related feelings such as exhaustion or failure, loss of inspiration, disengagement from employment, detachment from clients and colleagues and the appearance of wrong attitudes towards customers and themselves.

Although the characterisation of burnout has been subjected to numerous definitions, like the ones aforementioned, one of the core theories in this field is the one carried out by Maslach and Jackson in 1981. This study developed a multifaceted theory of the burnout syndrome that recognised three different dimensions within the experience of burnout: emotional exhaustion, cynicism and inefficacy (Maslach and Jackson, 1981). Furthermore, within this research, the Maslach Burnout Inventory was developed, which has become the most used burnout measurement instrument created to date, still prevalent in our time. This multidimensional theory of burnout allowed research to interconnect the three different scales with possible causes and consequences of its development. Due to the importance of this theory, an in-depth examination of each dimension will be carried out.

2.1.1 Emotional Exhaustion

Maslach and Leiter (2007) referred to this scale of burnout as the feelings of being overstretched and drained of the person's physical and emotional resources; employees experiencing this form of mental fatigue feel overused and unable to refill the mental energy disposed on their occupation development. This component of burnout identifies with the individual stress sphere of the syndrome. Emotional exhaustion has been regarded in burnout literature as a critical element of the concept of burnout and the dimension most studied and measured within the three that compose the experience of burnout (Alsharif, 2020). Moreover, emotional exhaustion has been strongly linked with work-place environments with high levels of stress and the service providers' over-involvement in their work (Lee *et al*, 2011).

2.1.2 Cynicism

The second dimension, also known as depersonalisation, corresponds to the relational component of burnout, which produces hostile feelings and extreme detachment towards a person's employment, leading to a loss of idealism and others' dehumanisation (Maslach and Leiter, 2016; Maslach *et al*, 2001). Cynicism is displayed as cold and distant attitudes towards customers and colleagues, limiting personal involvement, as a form of coping mechanism derived from the perceived low control over certain aspects of their work, such as job demands, role ambiguity and role conflict (Hollet-Haudebert, Mulki and Fournier, 2011; Schaufeli *et al*, 2002). Visible signs of cynicism encompass the use of pejorative language, the overuse of jargon, detachment from one's duties through longer breaks, and the over-rationalisation of situations; and these attitudes seem to stem from conflict in the workplace, the impossibility to partake in the decision-making process and the lack of fairness in the organisation (Totawar and Nambudiri, 2012). Furthermore, the development of cynicism in the workplace has been connected to unfavourable outcomes for the organisation, such as disengagement, low performance, and an increment of turnover intentions (Çına, Karcıoğlu and Aslan, 2014)

2.1.3 Inefficacy

The last dimension of burnout, also referred to as reduced personal accomplishment, corresponds with the self-assessment component of burnout, in which the person develops a tendency to devaluate himself and his capacities, generating feelings of inadequacy (Maslach and Leiter, 2016). Individuals experiencing this component of burnout feel that they are incapable of succeeding in developing their job, generating strong sentiments of incompetency. This dimension of burnout often originates from the previous dimensions, as feeling accomplished in one's employment is challenging when a person is depleted of energy and feels disconnected from work (Maslach and Leiter, 2008). The presence of feelings of inefficacy is often related to the absence of resources in the workplace to carry out the work properly, the lack of feedback, the level of autonomy in the position held, the level of clarity of the role carried out and the perceived support by one's peers (Valcour, 2016).

This multidimensional theoretical approach to burnout has not been homogeneously received by the scientific community in the research field, and there has been inevitable criticism about this model. An example of the detractors of Maslach's Burnout Inventory, henceforth MBI, is the Danish researchers Kristensen *et al* (2005). In their burnout study, these researchers elucidate that the MBI was created based on Maslach's elucidations of burnout and thus, this measurement system is not unbiased. Furthermore, these researchers believed that depersonalization is not a component of burnout but a coping strategy to deal with its effects. Moreover, the questions integrated into the MBI

were found by some participants in the study as difficult to understand or too restrictive. Finally, the MBI was developed with the prevalence that burnout only occurs in human-related services and professions, while the Danish investigators believed that burnout's ramifications could be extended to other professions. As a consequence of the reservations aforementioned, Kristensen *et al* (2005) developed a different burnout measurement instrument named Copenhagen Burnout Inventory, which positions the sentiments of fatigue and exhaustion at its core (Cox, Tisserand and Tariz, 2005). Similar to this inventory, other tools for the measurement of burnout have been developed since the conceptualization of this syndrome, such as the Oldenburg Burnout Inventory or the Single Item Burnout Measure, among others.

Nevertheless, the multidimensional model of burnout developed by Maslach and Jackson (1981) opened the field of study for different multifaceted burnout theories to be explored and developed and is still to date a reference in the field. Nevertheless, the research existent has not focus on the provision of homeless services, and it is in this base that this research will try to address this gap, analysing the different components of burnout aforementioned in the homeless service's providers

2.2 Stress

Stress has been regarded by Jin *et al* (2020) as one of the main components of the working life producing burnout. The existence of stress in the workplace is more widespread nowadays than in the past thirty years, as reported by the World Health Organisation (2007), henceforth WHO, becoming this subject a popular matter of research due to its impact on employees and organisations. The prevalence of stress has been identified as a result of contemporary working environments in which there is a higher pressure on efficiency and quality of work, a constant need to update employee's skills, and a fast-paced working rhythm (Sharma and Singh, 2016).

The process of stress occurs when events disturb the process of homeostasis, a self-regulatory process that the body naturally produces to preserve stability while adjusting to environmental changes generating physical and psychological disorders in the person experiencing it (Satish and Pestonjee, 2013). The definition of stress in the workplace presents specific difficulties as there is not a universal definition of this phenomenon but different approaches by different pieces of literature. The WHO (2007) has defined work-related stress as "a pattern of physiological, emotional, cognitive and behavioural reactions to some extremely taxing aspects of work content, work organization and work environment". Simultaneously, Robbins and Judge (2017) have defined occupational stress as a dynamic process in which a person is presented with an opportunity or request whose outcome is

regarded as ambiguous and important. Furthermore, work stress has been also defined as an emotional disorder that appears on an individual due to high environmental demands (Nisar and Rasheed, 2020).

Continuing with the elucidation of the phenomenon of work-related stress, research in this carried out by Robbins and Judge (2017) also suggest that there are two different sources of this type of stress, challenge stressors, such as workload and time constraints; and hindrance stressors, which are stressors that impede the person from reaching the completion of their tasks, such as lack of resources. Even though the research regarding these types of stressors is at early stages, there is evidence to support that challenge stressors are less harmful that hindrance stressors (Pearsall, Ellis and Stein, 2009; Podsakoff, LePine and LePine, 2007).

The consequences of occupational stress impact the individual and the organisational context in which they are carrying out their job. Work stress produces behaviours, such as absenteeism, low job satisfaction, decreased performance, health claims, accidents, and drug abuse (Granster and Rosen, 2013). Moreover, work stress also impacts the psyche of the person affected by this condition, with individuals displaying characteristics such as forgetfulness, hesitancy in decision making, and burnout (Ferdian *et al*, 2020). The consequences that the stress produce are of an integral nature, affecting different spheres of human life which has position this negative outcomes of demanding jobs as a main subject of discussion among researchers.

There is a high occurrence of occupational stress among human service providers, and this has been thoroughly documenting through the research carried out in fields such as health provision services, social work practice and educational context (Farmer, 2020; Marc, Dimény, and Bacter, 2019; Singh, and Jain, 2017). Specific stressors apply uniquely to the provision of human services, such as emotional dissonance, known as the need of service providers to mask their own feelings, and stressors derived from the client and customer relationship; an example of the latter is unrealistic expectations placed on the service provider, or targeting verbal aggression towards the worker (O'Brien and Linehan, 2019: Dollard *et al*, 2003). Unique stressors that make the provision of human service a challenging field for work development.

The incidence of stress in homeless services is a subject that has been underexplored in the literature existent. Identifying this gap in current research it is the interest of this research to analyse the perception of stress of homeless services' providers.

2.3 Antecedents of Burnout

The incidence and appearance of burnout is not a linear process but a continuum involving several factors that erode the well-being of the person experiencing them (Green *et al*, 2014). The factors that influence the appearance of burnout have been shared into two different types, personal characteristics and organisational aspects (Yildirim and Sait Dinc, 2019). Due to this research's nature, primarily focused on the human resources process, a brief description of the personal factors influencing burnout will be carried out, followed by an in-depth description of the burnout phenomenon's work dynamics.

Personal characteristics are those intrinsic to the person and are not influenced by the organisational context's policies and procedures. These include age, relationship status, educational level, and professional experience; their influence in burnout among different professions has been documented in different research (Kim, Shin, and Umbreit, 2007). While personal characteristics of an individual have been regarded as less influential in the production of burnout than their organisation counterparts, these factors influence the perception of work environments; and, therefore, influence burnout outcomes (Dyrbye and Shanafelt, 2016; Maslach and Leiter, 2008).

Contrariwise, organisational factors are the principal predictor of burnout, and thus a detailed description of these aspects will follow, with the special incidence of the factors affecting the organisation chosen for this research. To elucidate the different work-related factors that affect the appearance of burnout, the theoretical model of job demands-resources will be used, and therefore, a further explanation of this concept will follow henceforth.

2.4 Job Demands-Resources Model

The Job Demands- Resources model, henceforth JDR, was first elucidated by Evangelia Demerouti in the twenty-first century and since its first conceptualisation, it has been widely accepted and applied in the research field (Jin *et al*, 2020). The JDR theory expounds that there are several factors present in the working relationship that impact the production of stress and, therefore, burnout; these can be classified into two categories: job demands and job resources (Hatch *et al*, 2019; Demerouti *et al*, 2001). This theory identifies chronic job demands as a predictor of mental strain, whereas job resources act as a motivator producing positive outcomes such as engagement (Bakker and Demerouti, 2007). Furthermore, the JDR theory also analyses the interaction between the two forces, has also an impact on the production of burnout, finding that the presence of job resources can act as

a catalyser to reduce burnout and produce positive outcomes for organisations, such as engagement (Hu and Schaufeli, 2011).

Research carried out in this field by Bakker, Demerouti and Euwena (2005) in an educational institution with the aim to analyse the relationship between job resources and demand and the production of burnout; found that the existence of high demands and the provision of low resources had an impact in the generation of two components of burnout, emotional exhaustion and cynicism. However, no relation with the third component, inadequacy, was found.

As in any other theoretical framework, this model has not been exempt from criticism. The JDR model has been regarded as descriptive rather than explanatory and too inclusive to the point of lacking rigidness (Schaufeli and Taris, 2014). Furthermore, this model has also presented concerns regarding the delimitations of the concepts of job demand and resources and their perception as positive or negative by the employee (Borst, Kruyen and Lako, 2019).

Nonetheless, due to the extensive use of JDR in research regarding burnout, as aforementioned, and the links it has proven with its components, the JDR model has been applied to the research carried out, and a more comprehensive demarcation of the job resources and job demands will proceed henceforth. This theoretical framework will provide the conceptual base necessary to identify the organisational factors influencing burnout in the organisation chosen for the study carried out and their incidence in the burnout experienced.

2.4.1 Job Demands

Job demands are defined but Turek (2020, p.61) as "those physical, psychological, social or organisational aspects of the job that require either sustained physical or psychological (cognitive and emotional) effort or both and are, therefore, associated with certain physiological, and perhaps, psychological costs". Work demands can be objective or subjective, and the stress derived from them arises when there is a perceived imbalance between demands and the person's capability to meet them (Kar, and Suar, 2014; Ramonienė and Gorbatenko,2013). These organisational characteristics are of exceptional importance as it is understood that high levels of work demands are positively linked to an increase in the threat of burnout (Seljemo, Viksveen and Ree, 2020). The components of job demands are very diverse; nonetheless, the subjects that will be analysed comprehensively are workloads, role ambiguity, role conflict and emotional demand, due to its importance in the development of human services.

2.4.1.1 Workload

Workloads have been defined as the number of tasks a person in an organisational setting needs to complete in a specific timeframe. In environments whit high workloads, employees cannot complete their tasks due to lack of capacity or time constrictions, generating discontent in their service providers and organisation and positively impacting the appearance of burnout (Maidasari, Absah and Siahaan, 2018). Previous research in a healthcare setting has identified that working with clients with high needs require a significant workload, which leads to increased job stress and possible burnout (Kim *et al*, 2019). Equally, high workloads and time constrictions have been regarded as recurrent issue in different researches carried out in service providers of homeless services (Mette *et al*, 2020; Robelski *et al*, 2020).

2.4.1.2 Role Ambiguity and Role conflict

Role ambiguity has been defined as the perception that individuals do not have all the information necessary to carry out their tasks or work. At the same time, role conflict has also been demarcated as the feeling of unable to fulfil incompatible job demands and expectations, incapable to satisfy all parts involved in the work development (Celik, 2013; Onyemah, 2008). Regarding these two component, the non-clearness of a person's role and the expectations attached to it, could propitiate burnout due to the extra emotional investments they need to make to clarify their role (Panari *et al*, 2019).

As the ones aforementioned, issues with role delimitation have been regarded as subjects that affect the job performance of its sufferers and lead to burnout (Tunc and Kutanis, 2019).

2.4.1.3 Emotional Demands

Emotional demands have been defined as the work requirements that involve a constant sensitive effort during communication with customers (Stelmokienė, 2019). Emotionally demanding environments have been the cause of cognitive stress, with symptoms such as difficulty focusing or making decisions, memory retention, and thinking correctly (Elfering *et al*, 2017). Furthermore, previous research proposes that burnout is a consequence of prolonged exposure to emotional demands in the workplace (Xanthopoulou *et al*, 2007). Working with the homeless has certain particularities regarding the demand on its employee's psyche, as this is a highly vulnerable societal subgroup who often have experienced traumatic circumstances (Kidd *et al*, 2007). Moreover, Providers of homeless services have been identified as likely to experience emotional distress due to the extended exposure to their client's trauma (Wirth *et al*, 2019).

2.4.1.4 Working Hours

Working hours are the hours in which the development of one's work takes place. In the literature available shift work is the source of well-being concerns as it has been linked to an overall decrease of the person's well-being in the form of physical ailments and mental health problems (Hulsegge *et al*, 2020; James *et al*, 2017). Furthermore, previous research has also found that there is a correlation between long working hours and burnout, especially in professionals under fifty years of age (Hu, Chen and Cheng, 2014).

There has not been significant research carried out in the field of the homeless services in the Republic of Ireland regarding the job demands stated above; Therefore, it would be of interest studying how the workload, job clarity and ambiguity, emotional demands and working hours, affect the work development of homeless services' providers and their feelings of burnout.

2.4.2 Job Resources

Job resources are defined by Demerouti *et al* (2001, p.501) as "the physical, psychological, social or organisational aspects of the job that may do any of the following: be functional in achieving work goals; reduce job demands and the associated physiological and psychological costs; stimulate personal growth and development". Job resources are organisational tools that promote motivation and wellbeing among employees and are a *sine qua non* condition to achieve organisational goals (Schaufeli, 2017; Van de Tooren and De Jong, 2014). Moreover, contrary to the findings related to burnout and job demands, previous research has observed that the presence of job resources is negatively related to the development of burnout. (Gonzalez-Mulé, Kim and Ryu, 2020; Van Veldhoven *et al*, 2020; Xu and Payne, 2020). A conceptualisation of specific job resources that affect the provision of human services will be carried out hereunder.

2.4.2.1 Peer Support

Peer support has been defined as the extent to which employees count on their colleagues to assist and support them during their work development (Tafvelin, 2019; Caringi *et al*, 2017). Positive relations among colleague's act as a sheltering system, protecting the employee from stress and improving their wellbeing and creating resilience. Conversely, the lack of social support in the workplace can lead to burnout, as employees do not benefit from healthy work environments where to share their work experiences (Ho, 2016). Several studies in the social field have highlighted colleague support's beneficial factors on employees working in human services. Among its benefits are providing emotional control through informal conversations and acquiring new approaches to difficult working circumstances through team discussions (Cook *et al*, 2020; Mcfadden *et al*, 2019; Ruch, 2007).

2.4.2.2 Organisational Support

Organisational support relates to the individual's perception of how the organisation values their contributions and safeguard their welfare (Turek, 2020). When individuals recognise that their organisation supports them, they develop a positive attitude towards work, increasing their engagement and pursuing higher goals. Nonetheless, if individuals don't perceive support from their companies, they acquire passive work attitudes, reduce their productivity, and may exhibit burnout symptoms (Wang and Wang, 2020). The organisational support could be shown in different working scenarios, such as the perceived justice and fairness of the organisational processes, the support provided by managers, the presence of inducements, and the existence of good working conditions (Zeng *et al*, 2020). Furthermore, in the provision of human services, organisational support has been found to be the leading predictor of burnout and a critical source of job satisfaction (Anomneze *et al*, 2016).

2.4.2.3 Supervision

Supervision is the formal process of supporting, guiding, advising, and promoting growth in the employees; as well as a tool to encourage quality and wellbeing in the workplace (Marashi and Bani-Ardalani, 2017). As the supervision variety is quite wide in nature, the dissertation will focus on clinical supervision, as this is the type present in the service framing this study. Herbert and Caldwell (2015) elucidate that clinical supervision goes a step further than regular supervision as apart from addressing the performance of the employee the supervisor also acts as an advisor and counsellor.

Previous research in this field has found that supervision supports employees processing events that occurred in the development of their work, reducing the levels of stress and burnout, allowing the individual to be more effective, and help them develop tools to cope (Wallbank and Hatton, 2011; Wallbank and Robertson, 2008). Additionally, in the provision of human services, the establishment of supervision provides for a safe space to communicate complex characteristics of the work, which are inappropriate to share with other people, protecting the employee from stress, burnout and trauma (Wallbank and Hatton, 2011).

2.4.2.4 Pay

Remuneration is a powerful job resource that provides motivation in the workplace and diminishes the work's unfavourable characteristics and the influence of burnout; well-paid personnel are more likely to provide extra effort, have a better attitude towards their job, and support the development of valuable personal characteristics such as good self-esteem and more confidence in carrying out their duties (Salami and Ajitoni, 2016). Furthermore, employees that receive a remuneration for their work that they consider fair are more willing to accept adverse job demands such as high workloads (Hu, and Schaufeli, 2011).

The JDR theory has not being applied to the provision of homeless services in the Republic of Ireland, and thus it would be of interest how the different demands and resources present in the work relationship in this field affect the appearance of burnout.

2.5 Impact of Burnout

The Burnout syndrome holds great significance in the working relationship due to its detrimental effects on the organisation and the individual. When the stress suffered by the individual in the workplace continues over time, the person could develop burnout, impacting the employee's life, overall health, and the organisation in which they are employed (Salvagioni *et al*, 2017). For this subject's proposal, the description of burnout's consequences will be divided into two spheres: the impact that burnout has on the individual and its effect on the organisation. It is noteworthy to mention that due to the lack of research in the homeless services, other human provision services will be used for the analysis of the effects produced by burnout.

2.5.1 Individual Effects

The effects that burnout can produce in an individual are varied, affecting their physical and mental wellbeing due to continuous exposure to distressing situations. Previous research has found that burnout is linked with emotional disorders such as low self-esteem, anxiety, depression, guilt feelings, insomnia, reduced tolerance of frustration and overall psychological distress (Rudman *et al*, 2020; Yang and Hayes, 2020; Suñer-Soler *et al*, 2014). Furthermore, research in the healthcare field has established that burnout is linked to severe cases of depression, and in some instances, suicide (Williams, Rathert and Buttigieg, 2020). Moreover, international research has found that the experience of burnout in the workplace can have an impact on the overall health of the individual at

a physical level, in the form of different ailments such as addiction to drugs or alcohol, musculoskeletal pain, cardiovascular disease, obesity and fatigue (Salvagioni *et al*, 2017; Shirom, 2009). Nevertheless, burnout affects not only the health dimension of an individual but also its relations and their overall quality of life; research illustrates that adverse sentiments generated by a person's job can have a spillover effect in their personal life, creating negative behaviours at home and subsequently damaging personal relationships outside of the working sphere (Neff and Karney, 2007).

2.5.2 Organisational Effects

The effects of burnout are not exclusive to the individual experiencing it, as it also affects the organisation in which they carry out their occupation. At an organisational level, personnel experiencing burnout present job dissatisfaction sentiments, are less productive and innovative, are more likely to be absent or avail of sick leaves, have low morale, produce more mistakes and injuries at work, present a reduction of the quality of their job and are more likely to show intentions to terminate their employment. (Gabriel and Aguinis, 2021; Salvagioni *et al*, 2017). Moreover, employees experiencing burnout tend to be less supportive of their colleagues and present a disregard for their organisation and its aims (McCormack and Cotter, 2013). These conditions produce a double threat for organisations as if the burnout individual leaves, the organisations lose a valuable resource to achieve their performance and productivity goals. Conversely, if the burnout employee stays in the organisation, it can cause additional costs for the company due to increased absenteeism, low performance, and diminished quality (Kutz, 2017).

Considering the numerous adverse effects that the burnout syndrome causes and the loses it generates, not only financially but personally, it is clear the importance of this syndrome in the working relation nowadays (Mehrad, 2019).

2.6 Provision of human services

The concern regarding the wellbeing in the provision of human services has been recorded in literature for several decades. This professional group aims to improve their customer's quality of life through interventions tailored to their needs and to achieve this, working in close, continuous contact with their customers is needed; positioning these working individuals at a greater risk of burnout (Dice, Simmons and Wolfenden, 2018). The roles in the human service provision and varied, which different age groups and backgrounds; these can include children and families, the homeless, the elderly, people with addictions, people with mental health issues, immigrants and people with disabilities, among many others (Moffat, 2011). It is noteworthy to express that burnout has the potential to affect any professions due to existence of stress in every professional field. Nevertheless, the existence of burnout in the provision of human services has been proven to be particularly high, as previous research illustrates (Sumner and Kinsella, 2021; Rasheed-Karim, 2020; Louison Vang *et al*, 2020; Chung and Choo, 2019; Li, Li and Castaño, 2019; Medina, and Beyebach, 2014).

2.6.1 Homeless Services

Copious amounts of research have been carried out in the provision of human services (Novyk and Mazur, 2021; Dall'Ora et al, 2020; Wu, 2020; Kheswa, 2019; Cooper et al, 2016; Ghanizadeh and Jahedizadeh, 2015); nevertheless, working with the homeless population presents its particularities that differ from other human services categories. Moreover, the working relationship with this subgroup of society and its impact on the service provider, such as burnout, has not been extensively researched. Nonetheless, the literature available provides a clear picture of what it entails to work with the homeless.

Homelessness is a complex phenomenon that can be prompted by individual factors, such as drug addiction and mental health; and structural factors as the lack of affordable housing or the lack of employment for low skilled individuals (Nagy-Borsy et al, 2021). Furthermore, a high percentage of the homeless population has been exposed to different forms of trauma, including sexual abuse, domestic violence, drug addiction, alcoholism, neglect and psychological abuse (Hopper, Bassuk and Olivet, 2010). Providing an extremely challenging scenario for homeless service providers to develop their practice.

The providers of homeless services are frontline workers caring for one of the most vulnerable groups of society, in high intensity and emotionally consuming environments, continuously exposed to traumatic experiences that could harm their emotional stability (Lemieux and Taylor, 2019). Through their work development homeless service providers have to confront several barriers that no other providing human service has to face, such as the special needs of this group caused by trauma or health conditions, managing multiple simultaneous needs and managing negative public attitudes (Mullen and Leginski, 2010).

In the development of their practice, building relationships with customers is paramount, but it can also produce adverse outcomes as some employees might find it difficult to set up and maintain boundaries with their customers (Wirth et al, 2019). Moreover, the work carried out with this chaotic group takes place at a plodding pace and is likely to be subjected to setbacks. Leaving service providers with feelings of lack of accomplishment and disillusionment (Kidd et al, 2007). Furthermore, Service providers that join the homeless services workforce tend to be idealistic and have big goals and expectations, but the unique characteristics of this type of work, described thereto, and the continuous exposure to high levels of stress can lead to job dissatisfaction and burnout (Mullen and Leginski, 2010).

In addition, another factor affecting the work development of service providers are the organisations in which they are immersed. Often these employees, count on limited resources to carry out their work while attending a high workload, which adds an additional layer of stress to their practice (Lenzi et al, 2020).

There has not been any research related to the relation between homeless services and the burnout generated in the providers of these services in the Republic of Ireland. Nevertheless, similar research was conducted by researchers Waegemakers Schiff and Lane in 2019. In their investigation, Waegemakers Schiff and Lane (2019) study the different components affecting the work development of providers of homeless services across 23 different organisation in the Canadian territory. This study aimed to analyse if these professionals presented sentiments of burnout, vicarious traumatisation and post-traumatic stress disorder as a consequence of their employment; and the relationship between these three factors. The research, due to the large scope, applied a quantitative approach to the research, with the use of questionnaire that included two different scales; the Professional Quality of Life, which measures compassion fatigue and burnout, and the post-traumatic stress disorder check list. The findings of this research prove the existence of high levels of burnout within the group surveyed, and that the adverse impact of the work environment on homeless providers is greater than emergency responders. Furthermore, the researchers also noted in their conclusion that their study might not be generalizable, and thus, further research is needed (Waegemakers Schiff and Lane, 2019)

The research carried out in this field, as the one described thereto, highlights the harsh conditions under which service providers carry out their practice and the development of burnout as a consequence of them. Taking in consideration the factors described thereto and the lack of research in homeless services, the dissertation carried out aims to provide an insight into the conditions that homeless service providers experience in their practice.

2.7 Summary

The literature review carried out in this chapter indicates the complexity of the burnout syndrome, and its conceptualisation. Paying special attention to the demarcation of the concept elucidated by

Maslack and Jackson (1981), that identifies three components within the concept: emotional exhaustion, cynicism and inefficacy.

There are several components that generate the apparition of burnout sentiments, but it is clearly identified that stress takes a principal stand in the generation of this syndrome. Furthermore, both personal and organisational factors have been found to participate in developing this syndrome; nevertheless, research in this field has focalised on the later.

The organisational factors described in the development of burnout have been divided into two subgroups according to the demands and resource theory, dividing these into job demand and job resources. Within these two groups, the concepts analysed included: workload; the existence of role conflict and role ambiguity; the influence of the emotional demands attached the profession; the impact of the hours worked in the person's wellbeing; the perceived organisational and peer support; the effectiveness of supervision; and the influence of the remuneration received.

Previous research has shown interested in the burnout phenomenon and the concepts attached to it, as shown in this literature review. Nevertheless, this compilation of literature has also revealed the lack of research related to this syndrome in the homeless services, with the particular incidence in the lrish context. This research aims to provide a more comprehensive understanding of the state and causes of burnout in homeless service providers in Ireland, in relation to their perception of burnout and its components, the identification of stress in the workplace and the job demands and resources existent that might support the apparition of this syndrome. Thus, supporting to cover the gap existent in current literature.

2.8 Research Aim and Questions

The literature review aforementioned has provided for the theoretical framework to develop and conduct this research. This dissertation has been developed with the aim to explore in-depth the experiences of burnout of the providers of homeless services and the organisational factors inciting this phenomenon. As exemplified in the literature review, there is a gap in the literature addressing this issue in homeless related services; therefore, this research would like to provide an insight regarding this syndrome in the Irish homeless services.

The research question that has guided this research is:

'The exploration of burnout in homeless services through the experiences of service providers'.

29

The resultant research sub questions are listed below:

1. How are the providers of homeless services experiencing burnout?

This question aims to understand the different components of burnout, following the conception of this syndrome elucidated by Maslach and Jackson in (1981). Through this question the concepts of emotional exhaustion, cynicism and inefficacy will be explored in relation of the experiences of providers of homeless services (Maslach and Jackson 1981).

2. How do employees perceiving their work in terms of stress and its impact?

This questions aims to analyse the level of stress that employees recognize in their work development and the consequences it produces in their employment and personal life. Jin *et al* (2020) recognized stress in their research as the main component producing burnout, while Nisar and Rasheed (2020) also identified stress as the main component of burnout resulting from highly demanding workplaces. Taking in consideration these theories and the challenging scenario that homeless services provide, this question will support the understanding of the level of stress perceived by homeless services' employees.

3. Is the burnout experienced influenced by job demands and job resources?

The aim of this question is to analyses the different components of the job demands and job resources in relation to the JDR theory and it incidence in the generation of burnout. Demerouti *et al* (2001) expounds in their research that there are several factors present in the working experience that impact the production of stress and, therefore, burnout, and these can be classified into two categories: job demands and job resources. Following this theory, the job demands: workload, role ambiguity/conflict, emotional demands and working hours; and the job resources: workload, role ambiguity/conflict, emotional demands and working hours; will be analysed to identify if they are supporting the incidence of burnout in the providers of homeless services.

No explicit hypotheses were made; the researched carried out has an exploratory nature, which looks to understand a subject in-depth; thus, the formulation of hypotheses could have affected the research's conduction, becoming this biased. No formulating hypothesis allowed this research to be open to the genuine feelings of burnout within the researched group and draw conclusions without any researcher's constraints.

Chapter 3. Methodology

3. Introduction

This chapter of the dissertation will provide an elucidation of the research aim and the objectives that the dissertation aims to achieve. Moreover, an exploration of the research methods and approaches available will be carried out, with particular incidence on the research methods and instruments chosen for this dissertation and the rationale for their selection. Furthermore, this chapter will also address the limitations of the research and the ethical concerns attached to it.

3.1 Research Philosophy

The first component object of analysis within the methodology implemented in this study is the research paradigm. Guba and Lincoln (1994, p.105) define the concept of research philosophy as "the basic belief system or worldview that guides the investigator, not only in the choices of method but in ontologically and epistemologically fundamental ways". In other words, it provides for a specific view of the world for its holder. Implementing a research philosophy is essential when conducting research based on human behaviours, as it determines the research design and supports the reliability and generalisability of the study. Furthermore, when questioning what philosophical approach would be best suited for a research, the research philosophy's adoption will be determined by the nature of the reality to be explored and the researcher's preference (Kankam, 2019). There are four research philosophies derived from the researcher's views of the study process: positivism, critical realism, pragmatism and interpretivism (Saunders, Lewis and Thornhill, 2019). Consequently, an analysis of each of these philosophies will be carried out.

- <u>Positivism.</u> This research philosophy seeks to provide a clear and accurate view of the world. It relates to the stance of natural scientists, which looks to observe realities to produce law-like generalisations. This philosophy is based on the use of scientific methods to produce pure data. (Saunders *et al*, 2019).
- <u>Critical Realism</u>. This philosophic approach strives to understand phenomena by studying its underlying structures, believing that reality cannot be assessed through our observations alone (Al-Sharif, 2020). Furthermore, due to the focus of its study, often critical realism takes the form of historical analysis (Saunders *et al*, 2019).

- <u>Pragmatism</u>. This action-oriented philosophical approach looks for information with the aim of solving an existent problematic. Moreover, the pragmatic approach believes that knowledge cannot be extracted from individual believes, experiences or habits (Maarouf, 2019).
- Interpretivism. This philosophical method aims to understand and interpret realities through individuals or groups' experiences, as it believes that reality is founded on social constructions. Moreover, the interpretivistic approach values subjectivity and inclusivity, as it seeks to validate multiple interpretations of the reality object of study (Nguyen and Tran, 2015; Goldkuhl, 2012). Furthermore, this approach values the collection of qualitative data to define the uniqueness of particular situations, promoting participation and creating an in-depth study of phenomena (Chowdhury, 2014).

The philosophical approach that aligns with the research questions and objectives identified is the interpretivistic philosophy. This research philosophy supports the fundamental concept of the research carried out, which is to understand burnout through the experiences and opinions of people suffering this syndrome. Furthermore, the interpretivistic stand allows this research to have multiple views of the issue studied, which is paramount for this dissertation, and values every contribution.

3.2 Research Approach

There are two different reasoning approaches that research can adopt to achieve its purpose, inductive and deductive. The deductive approach in research starts with the formulation of generalisations, also known as hypotheses, and seeks to prove if these generalisations apply to the specific context studied (Hyde, 2000). Conversely, the inductive approach to research looks to work exclusively with participants' experiences, being these the study drivers. Furthermore, this research approach generally starts with the formulation of aims and objectives and seeks to achieve these through the research process (Liu, 2016).

The research conducted has been approached from an inductive perspective, as no preconceptualisation of the findings was made, in the form of hypotheses, in this dissertation's conceptualisation. This research has been guided by its aim and subsequent objectives, allowing for conclusion drawing from the findings obtained. The rationale for selecting this approach is being guided by the interpretative quality of the inductive process, which aligns with this research's aim: 'the exploration of sentiments of burnout in the providers of homeless services in Ireland'. Furthermore, the inductive research approach seeks to condense extensive pieces of information, establishing relations amongst them, creating a summary of findings, with the aim to develop a theory regarding the phenomenon studied (Thomas, 2006). This corresponds to the research carried, in which several pieces of information regarding burnout were gathered to find commonality amongst them, to provide a view of the burnout syndrome in homeless services through the findings of the research.

3.3 Research Design

In the design of a research, there are two distinct data collection methods to be considered, quantitative and qualitative. These information-gathering strategies act as the framework of studies and determine the tools to be applied.

Quantitative research refers to a data collection method in which the connection between variables is studied, applying numerical measures and subjected to statistical analysis (Edwards, 2020). This approach is generally linked to the positivist research philosophy due to the high structuration of the information gathered, and its development of analysis to prove theories is frequently related to a deductive approach to research (Saunders *et al*, 2019).

Conversely, qualitative research methods refer to a series of methods related to the acquisition, use, and interpretation of non-numerical data to develop theory within the topic of investigation (Quimby, 2012). This research category is defined by its holistic view of the world, where behaviour is bound to situational characteristics and their interpretation of the context, being its primary concern the understanding of people experiences (Gupta and Awasthy, 2015).

The researcher's choice of the data collection approach could follow two variants, a mono-method or a mixed-methods approach. A mixed-methods approach to data collection uses both quantitative and qualitative tools to analyse the phenomenon object of study. This method has been considered more comprehensive than using only one method of information collection and provides more robust research findings due to the neutralisation of their respective weaknesses among qualitative and quantitative processes (Bracio and Szarucki, 2020).

In contrast, a mono-method approach to the research design is the implementation of only one of the methods of research aforementioned. Guest *et al* (2013) elucidates that mono-methods research approaches can adequately answer research questions and expounds that sometimes designing complicated research structures with multiple methods are not reasonable.

Considering the different approaches to data collections, the nature of the phenomenon object of study, and the time constrictions of this research, a mono-method, qualitative research was chosen and carried out. The qualitative concern regarding human experiences and the environment they are immersed in aligns with the research's aim and objectives projected. Furthermore, the application of only one method allows the researcher to focus on carrying out an in-depth exploration of the participants' feelings and experiences regarding burnout, especially under the time constriction of this academic research.

3.4 Research Tool

Qualitative research, as the one carried out, provides a diverse selection of research tools for the investigation of phenomena. The most common research tools for qualitative studies include observation, focus groups and interviews. All these inquiry methods are designed to allow the participants to express their thoughts in their own accord, with some of them, such as focus groups and interviews, providing a greater degree of freedom of response (Walle, 2015). Considering the literature related to qualitative research methods, the instrument deemed more appropriate for this research and thus implemented was semi-structured interviews.

In semi-structured interviews, the researcher formulates a list of prearranged questions that will unfold in conversation with the person interviewed. This research method allows people to express their ideas and opinions about a specific topic in their own concurrence (Qu and Dumay, 2011). Furthermore, this method has been found to produce robust data regarding people's experiences, opinions and perceptions, providing a calming atmosphere in which people feel more comfortable responding to questions (Queriós, Faria and Almeida, 2010). These features of semi-structured interviews are paramount for the research carried as it was a condition sine qua non, to explore the concept of burnout through the participant's experiences in a space in which they felt comfortable to share their insight.

The semi-structured interview carried out was divided into three distinct segments, related to the three research sub questions elucidated thereto (see appendix A). The first part was composed of a series of simple questions to break the ice regarding the interviewees' demographics, as well as their job description, longevity in the company and the employment field, and their motivation to work in this sector. The second section of the interview was based on the concept of burnout and its three dynamics. In this segment of the interview, the participants were asked about their conception of burnout, followed by three sets of questions regarding emotional exhaustion, depersonalisation and personal accomplishment. Lastly, the last portion of the interview was structured around a series of

questions related to the job demand and the job resources present, or not, in the organisation. Furthermore, after the planned questions were asked, the participants were provided with the space to ask further questions or provide further inside of events not included by the prearranged questions. The complete list of questions asked during the interview can be seen in appendix A included at the end of this dissertation.

During the interviews, the participants were allowed to answer the questions at their own pace and in their own accord; and only little prompting was provided to generate clarification or examples of the circumstances described. Moreover, to safeguard the interviewees' comfort, the participants were reminded at the beginning of the process that they could refuse to answer questions they did not feel comfortable with and that they could withdraw from the process at any time given. Nevertheless, none of the participants made use of these specifications in the development of the interviews.

The interviews were conducted through video conference on Microsoft Teams and recorded through the same program. Subsequently, the interviews were transcribed, and the information provided was analysed to find common themes and patterns.

Previous research has been carried out regarding burnout in the provision of human services following the same research design as the one carried out, qualitative research through semi-structured interviews, such as the studies carried out by Butler *et al* (2018) and Hoffman, Palladino and Barnett (2007).

3.5 Sample

Sampling has been regarded as the process of selecting a portion of a large group for a determined research purpose. Moreover, sampling provides a cost-efficient process to obtain accurate, faster results; and it has been viewed as the best option when resources are limited, as this research is (Bhardwaj, 2019). There are two distinctive sampling processes, probability sampling and non-probability sampling. In probability sampling, there is an equal chance for every member of the population to be selected. Conversely, in non-probability sampling, the probability of being selected does not apply to the whole population (Taherdoost, 2016). The research carried out seeks information from a particular group of individuals, the providers of homeless services experiencing burnout; therefore, a process based on non-probability sampling has been selected. The sampling approach implemented in this study is a mixture of 2 sampling processes convenience sampling and purposive sampling.

The first sampling technique applied was convenience sampling, which is defined as a sample in which the researcher interviews whoever is available and willing to participate (Walle, 2015). This form of selection is the most appropriate due to the personal nature of the circumstances to be discussed within the research interview and the expectation that not everybody would feel comfortable participating. Thus, a sampling process in which the person decides willingly to participate seems the most suitable for this research and, consequently, applied to the study carried out.

An email seeking participants was forwarded by the researcher to all employees in Focus Ireland who interacts with the homeless population in their working development; and this was facilitated by Focus Ireland's human resources team who share the employee address book with the researcher. The email counted with a personal presentation of the researcher, a statement of the research's aims and objectives and the expected involvement sought out. Additionally, the email also provided a participant information sheet in which more information regarding the research and the participation required was provided (documents available in appendix B and C).

The demand for participation gathered through convenience sampling was more extensive than estimated, with twenty-two employees stating their willingness to participate. Due to the great demand for participation, and considering the time restrictions for this research to be carried out, interviewing twenty-two employees was not feasible. Therefore, a second tier of sampling was applied, this time through purposive sampling. This sampling method has been defined as a method in which the participants are selected due to certain qualities they hold (Bhardwaj, 2019).

Within the twenty-two individuals willing to participate, a selection of twelve participants was made, considering the gender proportion in the organisation, and to include as many categories of services within the organisation as possible. It is noteworthy to mention that in the totally of services provided by Focus Ireland, the majority of these are focused on advice and information and case management with working hours within 9 to 5, therefore its proportion in the sample is higher. Equally, regardless of the presence of the organisation in different counties the biggest proportion of service is placed in Dublin, therefore the proportion of Dublin based interviewees is higher. A breakdown of the participants' profile can be seen in the table below.

| Participant | Gender | Type of service | Location | Working hours | Length of employment |
|----------------|--------|-----------------|-----------|------------------|-------------------------|
| Interviewee 1 | Female | Drop in service | Dublin | 9-5 | 1 year |
| Interviewee 2 | Female | AI/ CM | Dublin | 9-5 | 3.5 years |
| Interviewee 3 | Male | AI/ CM | Sligo | 9-5 | 15 years |
| Interviewee 4 | Male | СМ | Limerick | 9-5 | 1.5 years |
| Interviewee 5 | Female | AI / CM | Dublin | 9-5 | 3 years |
| Interviewee 6 | Male | AI | Kilkenny | 9-5 | 2 years |
| Interviewee 7 | Female | Residential | Dublin | Shift work | 3.5 years |
| Interviewee 8 | Female | AI | Dublin | 9-5 | 3 years |
| Interviewee 9 | Male | Residential | Dublin | Shift work | 1 year |
| Interviewee 10 | Female | Drop in service | Dublin | Shift work | 3 years |
| Interviewee 11 | Female | Relief worker | Dublin | Shift work | 9 months |
| Interviewee 12 | Female | СМ | Waterford | 9-5 | 4.5 years |

3.6 Pilot Study

Prior interviewing the participants identified, a pilot study was carried out. A pilot study is a process utilised to prove the study's viability and improve the research quality, and can be carried out in two different forms: feasibility studies and pre-tasting pilot studies. Feasibility pilots are smaller versions of the research to be carried out, while pre-testing pilot studies seek to examine a determined research instrument (Malmqvist *et al*, 2019). The pilot study carried out identifies with the pre-tenting kind and has been conducted to test the viability of the interview's question, the time for its completion and identify possible faults.

A person not included in the 14 participants of the research was selected from the pool of 22 individuals willing to participate identified. The pilot study's completion indicated that the questions were easily understandable and provided enough time for personal elucidation. The time frame intended was achieved, and no faults were identified. Nevertheless, the pilot study helped the

researcher build up the confidence to carry out the interview process and feel comfortable within the role carried out.

Additionally, due to the willingness to participate received by a few project leaders and team leaders, which harbours a managerial position within the organisation, a pilot was carried out with a team leader to assess the interview's feasibility with this subgroup. The pilot study demonstrated that the interview questions did not apply to the employees in managerial positions as they are not in close contact with service users and have higher control of the work demands and resources than regular employees. Therefore, the participation of team leaders and project leaders of the organisation has not been included in the research.

3.7 Analysis of Data

Twelve interviews were carried using Microsoft teams' platform and recorded with the consent of its participants. The qualitative data obtained from said interviews were transcribed and analysed using a thematic analysis. This type of information analysis examines several events under the same core concept to support the development and analysis of cause-effect theories that support the understanding of phenomena (Turan, 2020).

The 12 interviewees carry out their work development in various types of homeless support services and in different areas of the Irish geography. The connection amongst the participants in this research is the experience of burnout and their employment within the same organisation in the provision of homeless services; Thus a thematic analysis was deemed the best option to analyse the qualitative data obtain to identified common themes and develop a theory regarding burnout in the homeless services.

The thematic analysis of the qualitative information obtained follows the theoretical framework developed by Braun and Clark in 2006. Their research developed a guide on how to carry out a thematic analysis identifying patterns and creating interconnections and providing clear guidelines for its rigorous conduction (Braun and Clark, 2006). This approach to qualitative analysis has six phases that were applied to the research carried out, and that will be expounded henceforth:

 <u>Data familiarisation</u>: This section of the analysis correlates with the multiple readings of the information gathered in an active manner, immersing the researcher in the material sourced to identify themes or ideas. In this regard, the interviews' transcriptions were read multiple times to identify the main ideas in each of them.

- 2. <u>Generation of codes</u>: Coding relates to the most basic idea within the themes identified. In this phase, the information analysis takes the form of the coding process of the initial ideas identified, providing an equal amount of attention to each part of the information gathered. Coding can take different forms; nevertheless, in the analysis carried out, the coding process utilised different coloured highlighters to code each of the interviews sourced to identify the ideas found visually. Furthermore, notes were also made in the transcriptions gathered to clarify the codes found.
- 3. <u>Theme searching</u>: This section of the analysis focuses on the analysis of the codes recognised, to developed interconnecting themes, which broader constructions of the concepts previously located. Furthermore, in the identification of these themes, a correlation between them is also made, identifying the main themes and the subthemes. In the research carried out, the identification of themes took the form of a map, in which a description of the codes and the subsequent themes and subthemes were elucidated.
- 4. <u>Theme review</u>: This phase of the data review involves reviewing all the themes identified to refine them. In this section, some themes will be disregarded as there is not enough data to support them, and others themes of broader implications can get divided into various smaller, more specific themes. Furthermore, at this stage, the whole data gathered must be reviewed to verify that the theme map correlates to the information obtained and to identify any valuable information missed. In the study conducted, the theme map was reviewed and reorganised in relation to the totality of the information obtained, and the transcription of the interviews was read an additional time to identify any critical information overlooked.
- 5. <u>Defining and naming themes</u>: This part of the analysis focuses on the themes identified, brooding its conceptualisation and refining its content. The themes are correlated to extracts of the data and organised coherently, providing a written analysis of each of them. Furthermore, after this phase is completed, the themes would be clearly defined. For this purpose, a list of the themes identified in the theme map aforementioned was carried out, providing specifications for each theme related to the information obtained.

6. <u>Producing the report</u>: The final section of the thematic analysis corresponds with the development of an in-depth analysis of the themes identified and creating a report. Here, the complex information is narrated compellingly, convincing the reader of the analysis's quality and legitimacy. Moreover, the analysis should follow an accurate, concise, rational, non-repetitive and appealing structure. The report of the findings provided by this research will be expounded in Chapter 5 of this dissertation.

(Braun and Clark, 2006).

This qualitative data analysis model is a straightforward and flexible tool to examine research without technical knowledge. Nevertheless, like other approaches to the analysis of information, the thematic analysis of qualitative research has its downsides. This type of analysis could be easily biased by the researcher's own thoughts, damaging the legitimacy of the analysis. Moreover, this analysis is a summary of the data gathered and a process of rationalising it, and thus it should inform the reader of what the information means, not just summarising. Furthermore, the researcher has to consciously safeguard the congruence between analysis and theoretical framework to provide a good analysis (Javadi and Zarea, 2016).

3.8 Ethical Considerations

Research ethics are a subdivision of applied ethics that provides guidelines for researchers to conduct studies, safeguarding the participants' wellbeing and the research's veracity (Akaranga and Makau, 2016). The protection of the subjects involved through the application of ethical principles has been recognised in all types of research. Nevertheless, in qualitative research, ethical considerations take a robust stance due to the deepness of the processes used, as distress can be caused by exploring one's emotions or recalling past events (Binti Mohd Arifin, 2018).

In designing the research carried out, the sensitivity of the study's topic has guided its conceptualisation. It can be extremely challenging for people to express their sentiments and opinions about such a personal subject as burnout in homeless services, and some people might also fear being judged. Furthermore, the setting of the burnout sentiments, in this case in the workplace, add an extra layer of discomfort due to the essential nature of an individual's employment to survive in modern

society. Therefore, the ethical concerns of this research have been considered through its whole development.

3.8.1 Pre- and During the interview

Before the conduction of this research, an outline of the ethical considerations and process to safeguard these was approved by the NCI's ethics committee. Equally, while conducting the research, several steps were followed to safeguard ethical compliance.

All employees in the organisation received an information sheet (appendix C), as aforementioned, providing an overview of the research, its rationale, and the expected participation to ensure informed participation. Furthermore, the information contained in this document emphasised that participation in this research is entirely voluntary and that the participants can withdraw from the study at any point.

The employees that decided to take part in the research carried out were provided with a consent form (appendix D), outlining the aim of the research, the participation required, the risks associated with the research, the benefits of participating, request of the agreement for the use of video conference recording, the confidentiality and data security of the process, and the researcher's personal information in case any question arise. The consent forms had to be completed and returned before the interviews' conduction to safeguard that all participants were aware of the research's expectations and requirements.

During the interviews, the participants were informed once again about the possibility of withdrawal from the study; and additional consent for recording the interview was sought.

3.8.2 Data and Storage

The data collected has been encrypted and anonymised, with all identifiable information regarding the participants been removed. Furthermore, regarding the data acquired, the processes followed the current legislation regarding data protection. The data produced has been sorted with identification codes and placed in a password protected folder in the researcher's personal computer, to which no other person has access. This data will be deleted a year after the completion of this research.

3.8.3 Partakers

The interviews carried out took place through video conference using the platform Microsoft Teams, and their duration was around 45 minutes to 1 hour each. During the interviews, the research maintained a welcoming and pleasant environment to allow interviewees to feel comfortable with the process and generate authentic personal answers to the questions researched. The researcher

maintained eye contact, address the participants by name, and nonverbal and verbal queues were produced to show that active listening was maintained.

The questions elucidated were designed using simple wording for the easy understanding of the participants in the study. Moreover, the partakers were provided with opportunities to ask questions or clarification from the researcher if needed, and the possibility to refuse answer to any questions they felt uncomfortable with.

Due to the emotional content disclosed through the research, the participants were encouraged to seek mental health professionals' support if the research or research subject was affecting their wellbeing.

Furthermore, the day after completing each interview, the research provided each participant with a follow-up email to safeguard their wellbeing and to open a communication channel for any further questions the employees might have.

Great care was placed in this research design to avoid situations that might produce harmful effects on its participants.

Chapter 4. Findings

4. Introduction

This chapter of the dissertation will focus on analysing and discussing the findings obtained from the interview process' data. The findings originating from the thematic analysis described thereto have been divided into seven themes and the consecutive subthemes to support answering the research question, 'the exploration of burnout in homeless services through the experiences of service providers'; and the three research sub-questions identified thereto within chapter two. Quotes from the participants will accompany the analysis of each theme to exemplify the key points revealed through the analysis of the data.

4.1 Theme 1. Identification of Burnout

The definition of burnout was explored with all the participants. The answers received provided a wide range of definitions that differed from each other, but with the common characteristic that all of them identified burnout as being composed of multiple factors such as lack of motivation, a decline in physical health, poor mental health, diminished sense of care for one's work and a reduced sense of accomplishment among many others. All these components are present in the literature review as constituents of the experience of burnout and exemplify a good understanding of the interviewees' concept.

One concept present in most of the demarcations of burnout was the association of burnout with stress. Nine interviewees associated burnout with the existence of stress in the workplace and recognized its impact on their wellbeing

"my understanding of burnout is that it's kind of an emotional and physical response to stress from job stressors, that kind of affect your mental and physical health, and that it can present similar things to like depression or anxiety"- Interviewee 9.

All participants were asked to identify if they felt burnout, to which eleven participants recognized that they are feeling burnout to some extent or that they are in the process of getting burnout. The reminder partaker indicated that at times they feel burnout, as the feeling fluctuates depending on the workload and the responsibilities attached to it.

4.2 Theme 2. Emotional Exhaustion

Emotional exhaustion is one of the components of burnout analysed through the interview process, with the aim to answer the research sub-question 'how are the providers of homeless services experiencing burnout?'. All of the twelve participants recognised elements of this stage of burnout within their work practice, and an in-depth analysis of the emotional exhaustion's components will proceed in the following sub-themes identified.

4.2.1 Sub Theme. Physical and Psychological Fatigue

All interviewees disclosed that they often had experienced fatigue related to the work carried out; and that the presence of this fatigue was not necessarily attached to the business of their environment but the emotional content of their interactions.

"I come home from a regular shift not necessarily a longer one, just one that wasn't overly taxing, just a normal day, and I am absolutely exhausted for no reason" - Interviewee 9.

"there are days that the job just sucks the life out of you, and you don't really want to do anything after... I feel sometime we take on some many of the problems other people have that it feels like drowning mentally" - Interviewee 8.

4.2.2 Sub Theme. Energy Replenishment

Seven participants stated that the breaks provided did not feel enough to go back to work feeling replenished.

"I don't know if I feel rested, I don't feel renewed, I dread going back after the weekend I dread it, the weekend never feels long enough" - Interviewee 2.

Adittionally, two respondents acknowledged that sometimes they feel the time off provided is not enough, depending on the circumstances experienced at work, such as managing crisis which are stressful in nature. It is also significant to mention that three of the participants identified that their capacity to recharge their energy levels had been affected by the social and physical restrictions instated due to the Covid 19 pandemic. "I think with the pandemic, this tiredness has been heightened like honestly we don't have anything really to do on the weekends and with the 5km restrictions and everything... you just can't fully wipe your mind out because there is nothing to do to give you that spark. Like you can't see your friends, you can't go to the cinema, you can't go on holidays, all those things that made our job easier are aone" - Interviewee 6.

4.3 Theme 3. Cynicism

The interview carried out contained several questions regarding this component, intending to answer the research sub-question, 'how are the providers of homeless services experiencing burnout?'. All of the twelve participants provided accounts of circumstances that correspond with the experience of cynicism in the workplace, in the form of detachment from the organisation, treatment of customers in a distanced manner, or through the avail of additional breaks not provided by the organisation. All these components have been identified as subthemes.

4.3.1 Sub Theme. Detachment from Clients

When asked if they have ever treated clients in a depersonalized manner, without acknowledging the person's background, ten participants recognized that they had treated customers in this manner, whether this is done consciously or unconsciously, due to the stress and the hectic nature of the work carried out.

"I think when you are a bit overwhelmed with everything and like there's a lot going on in work and you get different people ringing in, you got like a long list of people to be calling back with different like a real range of different issues like I think sometimes yes, like you can just sort of treat them a little bit like another number that's ringing" - Interviewee 1.

"it is really hard because we are dealing with so many different stakeholders in our jobs, and we have to tick this box, and we have to record this and do that, and it is heavy, I suppose trying to be with a person wholeheartedly 100% it is diluted because of the heaviness, of the elements of the job" - Interviewee 12.

Four of the respondents that recognized the depersonalized treatment of their customers described this depersonalization as a coping mechanism or a learnt skill to deal with difficult situations in the

workplace to safeguard their personal wellbeing; not identifying this component as a negative impact of burnout but a skill necessary for their work development.

"There's times where I think it's actually something you have to develop in our job. You have to develop a kind of not a callous approach, but you have to develop an approach where you can cut through a lot of the b*******" - Interviewee 6.

4.3.2 Sub Theme. Organisational Engagement

Two of the interviewees expressed that they were engaged with the organisation, with one of them stating that he/she feels identified with the aim and values of the organisation. One participant stated that she felt partially engaged to the organisation through all employee emails and practices instated by the organisation, such as reflective practice or supervision. Nine participants expressed that they did not feel engaged with the organisation. Not a common reason was found in their disengagement expressions, which listed causes such as their job not being their end goal, having too much to do in their work development to feel engaged, and being physically detached from the primary organisational resources.

"I don't feel very connected to Focus Ireland. I don't feel overly, connected to the organisation as a whole" - Interviewee 2.

"In my service, in my team I would be, but in the organisation as a whole no. I think it is because it is a high pressure job that everyone is doing what they can to probably stay afloat" - Interviewee 11.

Two of the participants identified that even though they did not feel engaged with the organisation as a whole, they felt highly participative in the services where they carry out their job.

"I am really engaged with my fish ball area of the project I work and the immediate connecting projects, but that would be it"- Interviewee 9.

4.3.3 Sub Theme. Sick Days and Extra Breaks

The responses received were polarising with a division of six participants that have never requested a sick day for personal reasons and six other participants that have taken a sick day to provide a break from the extenuating circumstances at work. The reasons provided by respondents for the utilisation of sick days in this manner was related to a sense of being overwhelmed with work and the need for

space to recharge mentally. Four of the interviewees that have requested sick days for non-medical reasons stated that they consider their mental well-being as a valid reason to request a sick day, even if the organisation does not allow this in their sick policy; And that the organisation should explicitly recognise this.

"this is quite I suppose conflicting for me because I am very much of the view that mental health restoration I think it is really important, I would say that taking a day off for your mental health if you are not feeling 100% can be, could be ahm... you know regarded as a sick day" - Interviewee 12.

"it's one of those days when you can't get your head off the pillow you just think for today, I can't do this, but I wouldn't say that it's not a sick day. I would consider that as a sick day, and I think all employers have to recognize this, that if we don't do those things to protect ourselves, we won't function properly as a worker"- Interviewee 3.

Seven participants recognised that they often take longer breaks than those provided. The rationale provided for this behaviour was primarily a coping mechanism to deal with feelings of being overwhelmed with the work.

"I just think sometimes the thoughts of going back again are overwhelming, you're like oh Jesus, please don't let this lunch end haha, and you might just need a bit more time to just clear your head"- Interviewee 1.

Theme 4. Inefficacy

The exploration of this burnout component allows the research to answer one of the research subquestions proposed 'how are the providers of homeless services experiencing burnout?'. All interviewees had a positive perception of themselves in relation to their work. They all felt that they were carrying out their work correctly and identified themselves as good at their job.

"I think I am good at my job. I mean, I've always tried to give 100% and you know, we're not always able to give that, but I think I've done the best I could at all times" - Interviewee 2.

"I would think I am, yeah. This is awful, I am shining my halo. Like when I stand back and take the time to reflect, you know... I normally do that through supervision and reflective practice, I can recognize that I am good at my job" - Interviewee 12.

Eight participants recognized themselves as valuable for the services and teams where they develop their professional practice. The participants that recognized themselves as valuable found that they

have had good feedback from management or the peers working along with them, and some of them identified themselves as valuable assets due to the knowledge and skills acquired through their professional practice.

"For my team definitely, like you need to have technical knowledge to carry out my job and certain skills to understand legislation and policy. Also my manager is really good at giving feedback, so I feel that she recognises the work I do and that makes me feel valuable"- Interviewee 8.

"yeah I think I am valuable, ahm... I think I am a team player, if something motivates me to do my job is my team and not letting them down. mmm.. and I think I've learnt a lot doing this work that I do have skills and knowledge that is valuable because it could be tricky how to navigate all these different systems" - Interviewee 2.

4.4 Theme 5. Stress

Several questions regarding stress and its impact were examined during the interviews with the aim to answer the projected research sub-question 'how do employees perceive their work in terms of stress and its impact?'. All participants identified their job as stressful. The rationale for this perception was very varied in nature, being the main grounds, the fact that the service users are experiencing crisis and dealing with complex circumstances, the emotional load of the work carried out, and the job's unpredictability.

"I think the job that all people in Focus do is stressful because you know, as I say, you're dealing with really complex people with really complex lives" - Interviewee 3.

"It is stressful work, and that that's true. It's a blend of highly emotional work combined with a lot of responsibility" - Interviewee 4.

Eight of the participants recognized that the stress experimented during their work development affected how they carry their work. The working attitudes recognize developed by the participants included a diminished sense of care for the clients, the conscious avoidance of specific tasks, and the development of feelings of frustration towards other agencies or stakeholders.

"If there is a day that's particularly stressful, I would potentially avoid doing certain tasks, and because of that, like, there might be a day when I'm having a particularly stressful day, I just don't care as much, my head isn't in it" – Interviewee 6.

Eleven participants identified that the work setting's stress affects their personal lives, even if this happens unconsciously. The impact of work stress in the personal life took the form such as irritation towards cohabitants, emotional and physical tiredness and absenteeism from personal life.

"Yes, like as I said before it can put me on a mood, or it can make me extremely tired, or even sometimes a bit snappy to my partner"- Interviewee 8.

"I've had people being like. You're hardly ever here, you're very stressed when you come home and or that you become a bit of a robot when things are particularly bad. I think it definitely affects my personal life" - Interviewee 4.

Theme 6. Job Demands

The interviews carried out contained several questions regarding the resources present, or not, in the organisation with the aim to support answering the research sub-question, 'is the burnout experienced influenced by job demands or/and job resources?'. Within this framework, the participants identified several of the work demands assessed through the interview process as sources of detrimental effects on their wellbeing.

4.4.1 Sub Theme. Workload

All the answers given differed in nature as some services seemed to have a more hectic nature than others. Some components were present in most of the interviewee's accounts; most of the participants identified their workload as stressful, acknowledging that it fluctuates from day to day and that there is a high level of unpredictability attached to it.

"when you a sitting in a desk at an office where you have a good feel for what kind of workload is going to be there the next morning, for us that question mark is there from the moment you arrive until you go, you know... anything can change, so your workload changes and your stress as well" -Intervieew 9. "I think I always work at a high level of stress and I think a lot of that comes from the workload" -Interviewee 3.

Seven participants voiced that they are unable to complete all the tasks expected from them during the workday, having to prioritize and leave things uncompleted. From this group of respondents, 2 interviewees also identified that the presence of extra duties, such as meetings or group supervision, adds and an extra layer of pressure to their workload.

"I think I am at the stage that I just accept you can have a to do list, but your to do list is never going to be completed, ever"- Interviewee 12.

4.4.2 Sub Theme. Role Ambiguity and Role Conflict

Nine participants stated that their role was ambiguous and unclear and that sometimes they have to carry out duties that they feel they are unprepared to deal with. The majority of the participants that felt their role was unclear disclosed that they had a vague induction, or not induction at all, and an overall lack of training.

"When you first start, nothing is really explained to you, your role is not really explained to you that well, there is not proper training or induction in this specific role"- Interviewee 2.

"No, I think in words in my contract, the outline of my responsibilities and duties ahm... when my manager tells me to be wary of my boundaries and my role and that kind of thing, yes 100%, it is. But realistically, in practice, it is absolutely not" – Interviewee 12.

The participant that holds a relief position across different services indicated that her role depends on the person on shift with her assigning her duties, rather than the position itself. This adds a level of uncertainty as the employee has completely different duties to carry out within the same role on different shifts.

Eleven participants identified that the customers often have unrealistic expectations regarding their roles. The respondents recognized that clients could be quite demanding in their requests, cannot see the boundaries of their power or the work necessary to achieve outcomes.

"sometimes it's just you know, like you're trying to do your best and they're like, well, you're not doing your job" – Interviewee 7.

4.4.3 Sub Theme. Emotional Demands

The participants were asked to express how would they define working with the homeless. The definitions present in most of the respondents included the following terms *emotionally taxing*, *exhausting*, *volatile*, stressful and *draining*. All the participants recognized that it takes an emotional effort for them to carry out their work.

"it does require an emotional effort from us, from me, like you don't know what someone is going to disclose to you while you are talking to them, so something that looks like a quick phone call it end up being them telling you that they are experiencing domestic violence or that they have been sexually assaulted" - Interviewee 2.

"because the client group we work with you can have a young person, service user, client whatever, maybe spit in your face, assault you, something that can be very emotionally damaging and you have to continue being professional" Interviewee 9.

"We can't be housing support workers and just deal with housing, we are dealing with the traumas that people has gone through and the emotional outpouring from them, and that I feel it is traumatizing for us" – Interviewee 12.

4.4.4 Sub Theme. Working Hours

There was a clear division between the employees working set ours on a nine to five basis and the employees carrying out shift work. The eight participants that work set hours found themselves content with their working hours. The four participants that work shift work, with three of them also working during the night, defined their working hours as *tough*, *challenging* and *difficult*. They all also recognized that their work-life balance, motivation, and perceived well-being had been compromised due to the unsocial hours carried out.

"you get up early in the morning but you come home late at night, so all you do all day is your job. Sometimes I feel that when I am at work I don't have time to do anything else. The days that I am at work I have to write off doing anything else"- Interviewee 11.

"It's just really hard cause you are tired all the time, really hard to I suppose being in a good mood, really hard to be able to motivate myself"- Interviewee 9.

Theme 7. Job Resources

The participants were asked to answer several questions regarding the resources present, or not, in the organisation with the aim to support answering the research sub-question 'is the burnout experienced influenced by job demands or/and job resources?'. All the participants identified resources within the organisation that were not achieving their goal of supporting their work practice and development, and this will be discussed in an in-depth manner in the following subthemes henceforth.

4.4.5 Sub Theme. Peer Support

All the individuals interviewed reflected that they have a good relationship with their peers and felt supported by them. The rationale behind these answers was that working with a trusted team with which there is a good relationship makes the work easier, acting as a catalyst of employee wellbeing, and allows to share the thoughts about the work carried out and certain circumstances attached to it with a group that fully understands what the person is going experiencing.

"I know that I have their support when it is needed and I can carry out the tasks with a clean mind, not stressed, knowing that everything is going to be alright Also because I know they have my back I know that I won't be put in some weird dangerous situation, because they are going to be there"-Interviewee 11.

"if a client is wrecking my head, I can outright say that to my colleagues that I am sick of this person today. It just makes it a bit easier like, especially if you're working with people you get on well like... It makes the day go faster, like you know that they're going to be sharing the workload with you, so it just makes everything so much easier"- Interviewee 10.

4.4.6 Sub Theme. Organisational Support

Eleven participants recognised that they did not feel supported by the organisation. The answers provided regarding the perceived organisational support were varied in nature. Five interviewees identified a lack of communication regarding changes in the organisation and initiatives implemented and felt that the employees' feedback was inexistent or not taken into consideration. Two participants stated that the organisation doesn't provide enough stability regarding job security and the type of contract provided. Three employees allude this lack of support to the inexistence of career progression in the organisation and the insufficiency of the training provided. Additionally, three interviewees

provided additional grounds for the perceived lack of support, alluding this to the lack of breaks in some services, the exclusion of relief staff from the organisation's wellbeing program and the disregard for promotion policies.

"It is an illusion of involvement sometimes, for the sake of it. Sometimes I feel like what is the point of me saying anything because you are not going to do anything about it anyway"- Interviewee 2.

"even before covid, I just don't see see the scope for professional development within the company there just isn't really that opportunity there"- Interviewee 4.

"like they send out some emails like mindfulness whatever, I don't think this is taking care of your employees. You know? I don't think that is enough. Because everybody can be just like here is a link to go meditate or whatever, these are not actual actions, it is more like in my opinion like they want to have a clear conscious about it. Again I am not getting those emails because I am relief, so that is another thing" - Interviewee10.

"I have seen in a number of occasions, that ahm... that there has being vacancies that haven't been advertised ahm... that have been put aside for somebody onsite"- Interviewee 12.

When asked about the perceived support provided by the immediate manager, eleven participants identified their managers are good sources of support, and that they do as much as they can to achieve this. With only one person describing an unsupportive relationship with their line manager.

4.4.7 Sub Theme. Supervision

Ten of the participants disclosed that they have regular supervision, where they feel supported and heard, and rated their overall experience with scores varying from six to nine, out of ten.

"I feel safe to say whatever I want in supervision. For the most part, I don't feel like it's going to come back and bite me and so. with my manager, I kind of feel like I'm actually comfortable bringing up issues that I am uncomfortable with, telling her what I don't like about the organization" – Interviewee 6.

From this group, five participants stated that their supervisor being their direct manager has implications to the level of honesty that can be achieved in the supervisor. The respondents felt that managers also carrying out the role of supervisors created a conflict of interests, as the feelings expressed during supervision could affect how managers view them as employees.

"I think there is a weird element of supervision being with your manager, because how am I going to be discussing certain things? or like if you have an issue with your manager and maybe you are not ready to talk about it with them, you don't have really anywhere to go with that" - Interviewee 2.

"I work so closely with my supervisor; it's it is a bit awkward to say things" - Interviewee 1

Two participants rated the supervision as zero. One of them stated that the line manager did not provide him with the supervision required, having less than one a year when it is intended for supervision to happen every eight weeks. While the remainder partaker that rated the supervision with a 0, stated that supervision is not provided for relief staff in general, and even though she has requested to have it as a personal choice, there has been no answer from human resources or management.

4.4.8 Sub Theme. Pay

Eight participants disclosed that the remuneration received for the work conducted is not reflective of the effort they put into their roles. There was an overall acknowledgement that they work for a non-profit organisation, but regardless of this characteristic, they felt that the job's stress did not match their salary.

"I was thinking this is not worth it, I don't know why I'm doing this, and a lot of it was down to pay and I thought that if the pay was a little bit better, the job is still s*** sometimes but at least you'll be like, well, at least I'm getting well compensated" - Interviewee 6.

"I feel for how hard the things we see are and the whole emotional effort we need to make daily, no... we don't get paid enough" – Interviewee 8.

Furthermore, one participant identified that segments of the organisation that are not in close contact with the homeless population, such as human resources or fundraising, are paid equally or higher than frontline workers, finding this unfair.

In addition, another respondent disclosed that the problems with remuneration in the provision of homeless services is more of a systemic problem, and that better conditions for homeless services providers or care worker should be demanded.

"I think we're giving statutory bodies far too good of a deal, and I think it would be very easy for social sector organizations to come together and demand better for their employees, not just the service users"- Interviewee 5.

4.5 Limitations of the Study

The limitations of a study are those characteristics of the research, or the methodology used that influence the analysis and interpretation of the findings (Price and Murnan, 2004). Regarding the research design, there are certain limitations attached to the use of qualitative methods of analysis. The research quality is over-dependent on the researcher's skills and could be easily influenced by the researcher's biases. Furthermore, the researcher's presence in the data collection process could influence the subjects' response (Anderson, 2010). Moreover, qualitative studies have been found to be not representative of the broader population as their quantitative counterpart due to higher levels of uncertainty held in this type of research design (Atieno, 2009)

Adittionally, other drawbacks have been identified within the study carried out, such as time constrictions, which has affected the number of interviews carried out, and the inclusion of only one homeless provider in this study. The research carried out is based in only one organisation providing homeless services, and some of the issues discussed within this research and its findings might not be representative of other organisations in the country providing similar services.

Chapter 5. Discussion

In this chapter, the research findings will be presented, exploring their meaning comprehensively for the dissertation carried out, and the relation with the literature review previously presented.

5.1 Burnout Dimensions

The principal aim of this research was 'the exploration of burnout in homeless services through the experiences of service providers'.

The definition of burnout, in the literature review, is characterised by the presence of chronic interpersonal stress as the underlying force propitiating the origination of this syndrome (Maslach, 1978). The presence of stress as the main component of burnout was identified by all participants in the research. Moreover, when analysing this concept through the interview process, the respondents provided numerous definitions of burnout with different characteristics such as lack of motivation, exhaustion, poor mental health, diminished sense of care for one's work and a reduced sense of accomplishment, among many others. All the components aforementioned are mentioned in the different definitions of burnout presented in the literature review carried out, such as the perception of fatigue (Freuenberger, 1978), lack of energy (Edelwich and Brodsky, 1980) and the perceived disengagement from one's job (Perlman and Hartman, 1982). The diversity in the answers provided in the different variations in this syndrome's conceptualisation, which lacks a worldwide accepted classification.

The demarcation of burnout elucidated by Maslach and Jackson (1981), was implemented as a guide to analyse the different components of burnout: emotional exhaustion, cynicism and inefficacy.

5.1.1 Emotional Exhaustion

The first dimension of burnout analysed was emotional exhaustion, which Maslach and Leiter (2007) identify as the sense of being depleted of the psychological and physical resources, unable to replenish these resources. Within this dimension, two different components have been analysed the sentiments of exhaustion and the sense of energy replenishment by the employees. All participants disclosed that they have felt fatigued from work carried out due to the emotional content perceived in their work. When enquired regarding the replenishment of energy through breaks such as annual leave and weekends, the majority of participants also identified that the time out of work did not appear enough to provide the energy replenishment needed; with a notable incidence on this component of the crisis experienced at work and the impact of Covid 19 in providing the outlets necessary to recharge their energy levels.

The high incidence in responses within the emotional exhaustion dimension of burnout confirms this component's presence in the experience of this syndrome by service providers of homeless support in the organisation researched.

5.1.2 Cynicism

Cynicism is the second dimension of burnout that identifies with the generation of adverse feelings and a sense of disengagement towards a person's employment, producing a loss of idealism and the dehumanisation of others in the process (Maslach and Leiter, 2016; Maslach *et al*, 2001). Due to the constant contact with others in the development of their practice, the concept of depersonalisation was analysed, with almost all participants identifying depersonalising attitudes towards their clients, whether this was made consciously or unconsciously, due to the stressful situations experienced. Some participants identified this detachment from the clients as a coping mechanism, not a negative consequence of burnout. Finding that supports the research carried out by Kristensen *et al.* (2005), which identifies depersonalisation as a coping strategy to deal with burnout and not one of its components.

Within the dimension of cynicism, following Maslach and Leiter (2016), the employees' perceived engagement was also assessed, with the majority of the participants identifying themselves as not engaged with the organisation. This finding could be viewed as a component of cynicism or a consequence of the presence of this component of burnout; as previous research elucidates, the development of cynicism in the workplace can be associated with unfavourable outcomes for the organisation, such as disengagement, low performance, and an increment of turnover intentions (Çına, Karcıoğlu and Aslan, 2014)

Half of the respondents had availed of sick days when they were not physically ill and that the majority of the participants make use of longer breaks not provided by the organisation to cope with the overwhelming nature of the job. These findings support previous research in the field that identified the use of longer breaks as a mechanism to detach oneself from the work duties as a visible sign of cynicism (Totawar and Nambudiri, 2012).

The high incidence in responses within the cynicism dimension of burnout and the attitudes developed in relation to this concept, explained thereto, confirm this component's presence in the experience of burnout by providers on homeless services in the organisation focus of research.

5.1.3 Inefficacy

This component relates to the appearance of feelings of diminished accomplishment and a devaluated perception of the capacity to carry out one's job successfully (Maslach and Leiter, 2016). This dimension of burnout usually appears as a consequence of the previous two, emotional exhaustion and cynicism, as it is difficult to feel accomplished when feeling exhausted and disconnected from their employment (Maslach and Leiter, 2008). Conversely to the findings in previous research stated thereto, all the respondents in the study had a positive perception of their work. They identified themselves as good at their job, with most participants recognizing themselves as valuable for their teams.

The research cannot support the existence of this burnout dimension within the provision of homeless services. This diversion from previous research could be explained by the close contact with their customers, as their work outcomes are easily recognizable through the interventions they achieved, such as supporting a client getting out of homelessness. It would be of interest to study this component of burnout more comprehensively in the provision of homeless services to study if the findings encountered regarding inefficacy are inherent to this type of services or not.

5.2 Stress

Stress has been regarded in the burnout literature as the main component producing this syndrome's appearance (Jin *et al*, 2020). And therefore, was of interest for this research to analyse the perceptions of work-related stress in homeless services providers, to answer the research sub-question 'how do employees perceive their work in terms of stress and its impact?'. All participants recognised their employment field as a highly stressful environment defined by the difficult circumstances they encounter in their daily practice, the emotional content attached to their work, and the job's unpredictability. This finding supports previous research that established that homeless services are high-intensity environments that can strain their providers' emotional stability due to continuous exposition to distressing circumstances (Lemieux and Taylor, 2019).

Due to the perceived stress experienced, most participants developed adverse attitudes towards their job, such as diminished care towards their clients, the avoidance of their tasks, and frustration towards other agencies they work in conjunction with. The attitudes developed from the stressed suffered aforementioned could be linked to visible signs of burnout as the development of a reduced sense of care towards clients and the avoidance of tasks or duties are signs of the presence of the second

58

component of burnout cynicism (Totawar and Nambudiri, 2012; Hollet-Haudebert *et al*, 2011; Schaufeli *et al*, 2002).

The overall presence of high levels of stress, like the one stated by the participants, and the overinvolvement in one's person job, as the homeless services require, has been positively linked with the development of the first dimension of burnout, emotional exhaustion (Lee *et al*, 2011). It could be understood that the high level of stress present in the professional development of the homeless services providers participating in this research is impacting the apparition of experiences of burnout within the dimension of emotional exhaustion and cynicism.

5.3 Job Demands and Job Resources

JDR theory elucidates, the demands of a job and the resources it puts in place have an impact on the generation of stress and the subsequent productions of burnout (Hatch *et al*, 2019; Demerouti *et al*, 2001). Therefore, it was of interest for this dissertation to analyse the job demands and resources present in the work development of the homeless services providers, to answer the research subquestion 'is the burnout experienced influenced by job demands and job resources?

The discussion of the findings provided by this dissertation in relation to the literature review existent will be divide into the two components of the JDR theory job demands and job resources.

5.3.1 Job Demands

Job demands are the organisational factors, whether physical o or social, that required a physical or psychological effort and that are associated with emotional costs (Turek, 2020); and it has been proven by previous research that high work demands influence the generation of burnout (Seljemo, *et al*, 2020). Therefore, it was an aim of this study, as aforementioned, to analyse if the demands of the job were affecting the wellbeing of the employees, and thus the generation of burnout. The job demands studied through the research carried out were workload, role ambiguity and role clarity, emotional demands and working hours, and an in-depth discussion of each of them will follow henceforth.

5.3.1.1 Workload

The interviewees provided different accounts for their perception of their workload, which can be understood by the different nature of the services they develop their practice in. Nevertheless, the majority of the participants disclosed that they perceived their workload as stressful due to the uncertainty and unpredictability attached to it. This unpredictability could be explained by the group they provide support for, which presents high demands due to the experience of crisis, such as evictions, sleeping rough, or drug addiction; working in a highly demanding environment, such as the one aforementioned, has been linked to an increment of the perceived stress by the employee and supports the generation of burnout (Kim *et al*, 2019).

Most of the participants disclosed that they rarely have time to complete all the tasks expected from them during their practice. This supports the findings in previous research that correlates the impossibility to complete all tasks assigned with highly demanding environments and the development of burnout. (Maidasari, 2018)

5.3.1.2 Role Ambiguity and Role Conflict

The overall perception of the majority of the participants in the study regarding the clarity of their role was that they felt it was not clearly demarcated, finding themselves carrying out duties that they felt they were not sufficiently trained for. Moreover, most of the respondents identified their induction process, which clarifies the job's general duties and procedures, as vague or inexistent. Thus, through these accounts, we could identify a high level of ambiguity in work carried out by homeless services providers.

On the other hand, regarding role conflict, almost all participants identified the customers as the primary source of the role conflict, as they sometimes expect more of their position or are incapable of seeing the extension and limitations of the work carried out by the employee.

The existence of situations where the role and the expectations attached to it are not clearly defined, like the one's described thereto, have been considered as generators of burnout; as the emotional effort a professional should put towards their job gets diverted to clarify the role carried out and its expectations (Panari *et al*, 2019). Thus the ambiguity and the unclear expectations of their roles could be influencing the sentiments of burnout existent.

5.3.1.3 Emotional Demands

The presence of these demands in the working relationship has been correlated to the generation of cognitive stress (Elfering *et al*, 2017), being the stress perceived the main initiator of burnout (Jin *et al*, 2020). All participants in this research identified working with the homeless as emotionally demanding, defining this type of work as emotionally taxing, traumatizing, stressful and draining. This

perception could be propitiated by the traumatic circumstances attached to this subgroup of society that has been often exposed to different types of trauma, such as drug abuse, domestic violence or sexual assault, among many others (Hopper *et al*, 2010).

Furthermore, the unanimous perception of this type of work as emotionally demanding supports previous research that states that prolonged exposure to emotionally demanding environments and traumatic events, such as the one's described thereto, produces the appearance of this syndrome (Xanthopoulou *et al*, 2007).

5.3.1.4 Working Hours

Concerning the participants' working hours, the most significant finding was identified in the group of employees carrying out long working hours through shift work, some of them even working during the night. The participants under unsocial hours recognized that their work-life balance had been compromised and perceiving low levels of motivation and well-being within themselves. These findings are consistent with the research existent in the field, which identifies shift work as the source of reducing employee's well-being (Hulsegge *et al*, 2020; James *et al*, 2017).

There are certain job demands present within the provision of homeless service in the organisation researched that are impacting the well-being of its employee, and thus, could affect the appearance of burnout. This supports the research previously elucidated in the literature review that positively correlates high levels of job demands with the generation of burnout (Seljemo *et al*, 2020).

5.3.2 Job Resources

Job resources are organisational factors that support goal achieving, decreases job demands, and reduces psychological and physiological costs while stimulating growth and development (Demerouti et al, 2001). The existence of job resource in the working relationship has been found to negatively impact the appearance of burnout, acting as a buffering system promoting wellbeing (Gonzalez-Mulé *et al*, 2020; Van Veldhoven et al, 2020; Xu and Payne, 2020); therefore, this research aimed to analyse if the job resources provided by the organisation were achieving their protective function.

The job resources that were examined were: peer support, organisational support, supervision, and pay, and an elucidation of the findings obtained about each one of these components will follow.

5.3.2.1 Peer Support

Regarding the support provided by their colleagues, all participants disclosed that they have a good relationship with their peers and feel supported. Additionally, the respondents expounded that relationship with their colleagues made the work more bearable, as they were able to share experiences with other individuals that understand the work carried out. This finding correlates to the theory that good working relationships promote well-being and resilience in the workplaces (Ho, 2016) and emotional support through team's discussions (Cook *et al*, 2020; Mcfadden *et al*, 2019; Ruch, 2007).

Conversely, burnout has been linked with unhealthy relationships among peers (Ho, 2016); this research cannot support this theory as the group interviewed did not present this situation.

5.3.2.2 Organisational Support

The organisational support was divided into two categories, the support perceived from direct managers and the support perceived by the organisation as a whole.

Regarding managerial support, almost all respondents identified that their direct managers were a good source of support and provided as much support as possible.

Conversely, when enquired regarding the perceived support from the organisation as a whole, the totality of the participants recognised that they feel unsupported by the organisation, due to a wide variety of reasons; which included lack of feedback, low job security, no possibilities for career progression, lack of standards of work, poor training, the exclusion of some employees from organisational practices, and the disregard for some policies and procedures. These statements support the existent theory in organisational support, which expounds that when employees perceive their employer as unsupportive, they acquire passive attitudes towards work and may display burnout signs (Wang and Wang, 2020).

Furthermore, like the aforementioned, feelings of unsupportiveness have been regarded in burnout literature as the main predictor of this syndrome (Anomneze et al, 2016).

5.3.2.3 Supervision

Almost the totality of the people interviewed for this research disclosed that they have regular supervision, in which they feel supported and listened to. This finding supports previous theory that found that supervision helps employees in processing situations related to their work practice, allowing them to be more effective and create strategies to cope (Wallbank and Hatton, 2011; Wallbank and Robertson, 2008).

Conversely, two partakers have different views regarding supervision, as for one, the process of supervision is inexistent, and the remainder one receives supervision sporadically. This disclosure, and the fact that supervisors are direct managers, causing restrictions in the level of sincerity displayed during supervision, could be affecting the appearance of burnout. As the protective space that supervision is supposed to provide to safeguard the employee from stress and burnout (Wallbank and Hatton, 2011), it is not provided or perceived.

5.3.2.4 Pay

The information obtained through the interviews regarding the remuneration received by employees for their work identified that most participants recognised that the payment received was not reflective of the effort their job requires. Creating a mismatch between the stress created generated by their work and the salary received. Moreover, one of the participants perceived unfairness in the organisation's pay scheme as frontline workers received an equal remuneration as the organisation's office-bound segments do.

This discontent with the remuneration received could be affecting the perceived burnout that the employees interviewed have; as previous research states, fair remuneration motivates employees and decreasing the influence of burnout and supports them coping with adverse job demands (Salami and Ajitoni, 2016). Therefore, as the remuneration received is perceived as inadequate, the sheltering component against burnout that it should provide is inexistent.

Chapter 6. Conclusion

This dissertation aimed to explore burnout in homeless services through service providers' experiences, providing an analysis of the different components affecting the impact of this syndrome. The literature review highlighted the complexity of this syndrome and the different elements that take part in the generation of this phenomenon and provided the body of knowledge needed to implement this study. Regarding the research conducted, this follows a qualitative approach using semi-structured interviews as the primary research instrument implemented. The interviews conducted were transcribed and examined using a thematic analysis of data.

The key findings obtained from the study will be expounded in reference to the three research subquestions that guided the research carried out.

- How are the providers of homeless services experiencing burnout?

The experiences of burnout mainly related to the dimensions of emotional exhaustion and cynicism, with all participants recognising behaviours or emotional outcomes related to these two dimensions. Regarding the third dimension of burnout, inefficacy, the research carried does not support the presence of this factor in the burnout developed by the providers of homeless services surveyed. Regardless of the challenging circumstances in which they carry out their work, all participants perceived themselves as good workers and valuable for their respective services.

- How do employees perceive their work in terms of stress and its impact?

This component was of extreme importance for exploring burnout, as stress has been regarded as the leading producer of the syndrome. The totality of interviewees identified their job as stressful due to their work's unpredictability, the crisis experienced by the service providers, and the emotional charge attached to their interactions with this subgroup of society. Confirming, therefore, the incidence of this component in the production of burnout. Furthermore, the research also showed that the stress experienced was correlated to the development of unfavourable working practices and impacted the participants' personal lives.

- Is the burnout experienced influenced by job demands and job resources?

Regarding these two components, the research found that job demands and resources influence detrimentally the work carried out in homeless services, consequently supporting the appearance of burnout.

Within the job demands present, most respondents identified their workload, role ambiguity and conflict, and emotional demands as sources of strain. Supporting the literature existent that expounds that this component's presence in the working relationship facilitates the generation of burnout. The last component analysed, working hours, was only deemed detrimental by the employees carrying work during unsocial hours. Supporting the literature existent that correlates shift work with a decline in well-being and the generation of burnout.

On the subject of job resources, the research analysed different resources present in their working development to identify if they were achieving their aim of supporting the development of well-being; therefore, impacting the stress levels suffered and the subsequent burnout. The majority of participants identified the organisational support and the remuneration received as the primary resources that were not achieving their goal of promoting welfare. Supervision was only perceived negatively by the participants who were not availing of it, or its provision was not applied appropriately, while the rest of the participants perceived it as a good source of support. Conversely, all participants' relationship with their peers was well perceived, identifying this as a promoter of well-being.

5.4 Suggestions for Future Research

The job resources and job demands analysed are related to only one organisation, as due to the time constrictions of this dissertation, as a more extensive scope in the research carried out was not possible. The findings in this research topic are bounded to the organisation chosen and their policies and procedure; therefore, the findings in this area might differ from one organisation to another. Therefore, it would be of interest for future studies to amplify the scope of research, including various organisations providing homeless services, to analyse if there is a commonality in the factors inciting the production of burnout.

The incidence of personal characteristics in the production of burnout was not analysed in the development on this research. This component in the production of burnout could be of importance for future research to assess its impact on burnout's overall production, apart from the organisational characteristics.

In the research conducted the representation of employees working shift work or night work was not very high as a small proportion of this type of workers requested participation. Therefore, it would benefit future research to try to achieve a more diverse group of participants, including more employees working unsocial hours. This research has explored the concept of burnout through the experiences of homeless services providers, providing an insight into a field that has been systematically understudied through the years. The provision of these services present similarities with highly stressed environments such as the healthcare and social work field, with not as much recognition. This research adds to the limited body of literature existent referred to homeless services and wished to promote an interest in this sector and further research in the field.

Chapter 7. CIPD Requirements

7.1 Recommendations

The researcher has established a series of recommendations arising from the findings encountered in completing this research, and they will be expounded hereafter.

1. Working from home after Covid 19

It has been palpable through his research that one of the primary resources for employees to bear the emotional complexity of their work is to make use of their personal time to do activities that help them unwind. Currently, most employees not placed in residential or drop-in services are working from home, reducing the time needed to commute to work, thus allowing for a more significant amount of personal time. It would be of interest for the organisation framing this research to continue this practice once the Covid restrictions are lifted. Instating a half and half teleworking and office working timetable could allow employees to achieve a better work-life balance while continuing their close relationship with their peer and conduct the duties that need to be carried out face to face with clients.

This recommendation would be cost-free and could be implemented with no temporality restrictions as all the resources for teleworking have been put in place at the moment due to the pandemic experienced.

2. The implementation of the 4-day work week

In the same light as the previous recommendation, due to the importance of personal time and the ability to recharge mentally to carry out a job with such emotional strain, the institution of a 4-day workweek could benefit the organisation. This change in the working hours could improve the employees' well-being and the overall perception of the organisation in terms of support.

This recommendation would be cost-free, and the time for its implementation should not exceed three months.

3. The review of the supervision policies

Due to the importance of supervision in the creation and maintenance of employee well-being, the policies and resources used at the moment should be revised. The relief staff should be included in the provision of supervision with a mandatory nature. This could support this type of employee in

feeling engaged with the organisation and provide them with the space to discuss work-related circumstances and their professional development.

Furthermore, the human resource management team should revise their policies regarding the quality of supervision, ensuring all employees are consistently accessing this resource every eight weeks.

This recommendation would be cost-free, and its implementation should not exceed two months, as all the resources are already in place.

4. Improve organisational communication

Improving the communication in the organisation providing two-way feedback improves the morale of the employee's feeling engaged. The organisation could benefit from creating systems or groups of discussion in which the opinion of employees is seriously taken into consideration. Furthermore, the organisation could also benefit from communication originating from the bottom up to discuss issues or ideas that primarily affect employees, such as the organisation's well-being agenda. Highlighting the need for all types of employees to be included in this communication.

5. The development of an in-depth training program

The review of the current training program and creating a more comprehensive training package could benefit the organisation greatly in terms of performance, while supporting employee satisfaction and commitment. The organisation should also include specific training for the different areas in the organisation, such as mental health, mediation, or housing law, apart from all employees' generic training.

The review and implementation of a full training program is estimated to take up to a year, and the costs of this implementation could vary, as some of the training could be developed in house, or in partnership with other homeless related centres or organisations.

7.2 Statement of Personal Development

I decided to course the Master in Human Resource Management, coming from a Social Work and Meditation background. Throughout my professional development, I had recognised that the manner in which organisations treat their employees, and the policies and procedures implemented, have a significant impact on the overall satisfaction that a person holds regarding their job. I believe this awareness and the knowledge acquired through my studies regarding how people behave and communicate and the need for a professional change gave me the foundation needed to further my education in this field.

The Masters' in Human Resource Management has been a great learning experience that has provided me with a whole world of knowledge that I did not have before and has expanded my horizons in terms of professional progression.

The dissertation carried out has been the most challenging and rewarding academic achievement I have accomplished to date, supporting me in improving my skills regarding time management and independent working and my overall knowledge of employee well-being and burnout.

One of the main difficulties faced during the completion of this dissertation was maintaining an equilibrium between my professional life, the completion of this dissertation and my personal life. This difficulty was easy to overcome after a set of strict deadlines were put in place, breaking down the work into small sections. In this regard, this dissertation has helped me develop my time management and independent working skills; skills, which will help my professional development move forward.

Apart from diving into a completely different sector, which was a challenge in itself, I had to face an additional challenge: my mother language. All my previous studies have been in my native language, and even though I was confident in my communication skills, I did not know if I would rise to the occasion. This dissertation has also helped me learn formal academic forms of communication, improving my overall confidence with my second language.

Furthermore, this dissertation has given me the possibility to research a subject that I am passionate about. Allowing me to enrich myself with the different views and notions regarding burnout and explore a current issue in homeless services that have been understudied and overlooked.

References

Akaranga, S. I. and Makau, B. K. (2016) 'Ethical considerations and their applications to research: A case of the University of Nairobi', *Journal of Educational Policy and Entrepreneurial Research*, 3(12), pp. 1-9. Semantic Scholar. Available at: <u>https://www.semanticscholar.org/paper/Ethical-</u> <u>considerations-and-their-applications-to-a-Akaranga-</u>

Makau/0aa01e9f5bf5cea523daf16693cfb9dde7096802 [Accessed: 4 April 2021].

Alsharif, A. (2020) 'The protective role of resilience in emotional exhaustion among dental students at clinical levels', *Psychology Research and Behavior Management*, 13(1), pp. 989-995. Directory of Open Access Journals. Available at:

http://search.ebscohost.com/login.aspx?direct=true&AuthType=ip,cookie,shib&db=edsdoj&AN=eds doj.f5111b41a1414081ad8a9433214abb43&site=eds-live&scope=site [Accessed: 8 March 2021].

Al-Sharif, R. (2020) 'Critical realism and attribution theory in qualitative research', *Qualitative Research in Organizations and Management: An International Journal*, 16(1), pp. 127-144. Emerald Insight. doi: 10.1108/QROM-04-2020-1919.

Anderson, C. (2010) 'Presenting and evaluating qualitative research', *American Journal of Pharmaceutical Education*, 74(8), pp. 1–7. Education Research Complete. doi: 10.5688/aj7408141.

Anomneze, E. A., Ugwu, D. I., Enwereuzor, I. K., Leonard, I., and Ugwu, L. I. (2016) 'Teachers' emotional labour and burnout: Does perceived organizational support matter?', *Asian Social Science*, 12(2), pp. 9-22. ReserachGate. doi: 10.5539/ass.v12n2p9

Atieno, O. (2009) 'An analysis of the strengths and limitation of qualitative and quantitative research paradigms' *Problems of Education in the 21st Century*, 13(1), pp.13- 18. Google Scholar. Available at: https://www.scientiasocialis.lt/pec/files/pdf/Atieno_Vol.13.pdf [Accessed: 04 April 2021]

Bakker, A. B. and Demerouti, E. (2007) 'The job demands-resources model: State of the art', *Journal of Managerial Psychology*, 22(3), pp. 309-328. Emerald Insight. doi: 10.1108/02683940710733115

Bakker, A. B., Demerouti, E. and Euwema, M. C. (2005) 'Job resources buffer the impact of job demands on burnout' *Journal of Occupational Health Psychology*, 10(2), pp. 170-180. Semantic Scholar. doi; 10.1037/1076-8998.10.2.170

Bhardwaj, P. (2019) 'Types of sampling in research', *Journal of the Practice of Cardiovascular Sciences*, 5(3), pp. 157-163. Directory of Open Access Journals. doi: 10.4103/jpcs.jpcs_62_19.

Binti Mohd Arifin, S. R. (2018) 'Ethical considerations in qualitative study', *International Journal of Care Scholars*, 1(2), pp. 30-33. ResearchGate. Available at:

https://www.researchgate.net/publication/328019725 Ethical Considerations in Qualitative Stud y [Accessed: 4 April 2021]

Borst, R. T., Kruyen, P. M. and Lako, C. J. (2019) 'Exploring the job demands–resources model of work engagement in government: Bringing in a psychological perspective', *Review of Public Personnel Administration*, 39(3), pp. 372-397. Sage Journals. doi: 10.1177/0734371X17729870

Bracio, K. and Szarucki, M. (2020) 'Mixed methods utilisation in innovation management research: A systematic literature review and meta-summary', *Journal of Risk and Financial Management*, 13(252), pp. 252-279. Directory of Open Access Journals. doi: 10.3390/jrfm13110252.

Braun, V. and Clarke, V. (2006) 'Using thematic analysis in psychology', *Qualitative Research in Psychology*, 3(2), pp. 77-101. Academic Search Complete. doi: 10.1191/1478088706qp063oa.

Butler, M., Savic, M., William Best, D., Manning, V., Mills, K. L. and Lubman, D. I. (2018) 'Wellbeing and coping strategies of alcohol and other drug therapeutic community workers: A qualitative study', *Therapeutic Communities: The International Journal of Therapeutic Communities*, 39(3), pp. 118-128. Emeral Insight. doi: 10.1108/TC-08-2017-0024.

Caringi, J.C., Hardiman, E. R., Weldon, P., Fletcher, S., Devlin, M. and Stanick, C. (2017) 'Secondary traumatic stress and licensed clinical social workers', *Traumatology*, 23(2), pp. 186-195. APA PsycArticles .doi: 10.1037/trm0000061.

Celik, K. (2013) 'The effect of role ambiguity and role conflict on performance of vice principals: The mediating role of burnout', *Eurasian Journal of Educational Research*, 1(51), pp. 195-213. ERIC. Available at:

http://search.ebscohost.com/login.aspx?direct=true&AuthType=ip,cookie,shib&db=eric&AN=EJ1059 818&site=eds-live&scope=site [Accessed: 12 March 2021].

Chowdhury, M. (2014) 'Interpretivism in aiding our understanding of the contemporary social world', *Open Journal of Philosophy*, 4(3), pp. 432-438. ResearchGate. doi: 10.4236/ojpp.2014.43047

Chung, Y. and Choo, H. (2019) 'A structural path to job satisfaction, burnout, and intent to leave among child protection workers: A South Korean study', *Children and Youth Services Review*, 100(1), pp. 304-312. Science Direct. doi: 10.1016/j.childyouth.2019.03.018. Çınar, O., Karcıoğlu, F. and Aslan, I. (2014) 'The relationships among organizational cynicism, job insecurity and turnover intention: A survey study in erzurum/turkey', *Procedia - Social and Behavioral Sciences*, 150(1), pp. 425-432. ResearchGate . doi: 150. 10.1016/j.sbspro.2014.09.045.

Cook, L. L., Zschomler, D., Biggart, L. and Carder, S. (2020) 'The team as a secure base revisited: Remote working and resilience among child and family social workers during COVID-19', *Journal of Children's Services*, 15(4), pp. 259-266. Complementary Index. Available at: http://search.ebscohost.com/login.aspx?direct=true&AuthType=ip,cookie,shib&db=edb&AN=14760

3334&site=eds-live&scope=site [Accessed: 13 March 2021].

Cooper, S., Carleton, H., Chamberlain, S., Cummings, G., Bambrick, W. and Estabrooks, Ca. (2016) 'Burnout in the nursing home health care aide: A systematic review', Burnout Research, 3(3), pp. 73-87. Science Direct. doi: 10.1016/j.burn.2016.06.003.

Cox, T., Tisserand, M. and Tariz, T. (2005) 'The conceptualization and measurement of burnout: Questions and directions', *Work and Stress*,19(3), pp. 187-191. Available at: <u>http://search.ebscohost.com/login.aspx?direct=true&AuthType=ip,cookie,shib&db=bsu&AN=18909</u> 027&site=eds-live&scope=site [Accessed: 9 March 2021].

Curtis L., Moriarty J. and Netten A. (2010) 'The expected working life of a social worker', *The British Journal of Social Work*, 40(5), pp. 1628-1643. ResearchGate. doi:10.1093/bjsw/bcp039

Dall'Ora, C., Ball, J., Reinius, M. and Griffiths, P. (2020) 'Burnout in nursing: a theoretical review', Human Resources for Health, 18(1), pp. 1-17. Directory of Open Access Journals. doi: 10.1186/s12960-020-00469-9.

Demerouti, E., Bakker, A.B., Nachreiner, F. and Schaufeli, W. (2001) 'The job demands–resources model of burnout', *The Journal of applied psychology*, 86(1), pp. 499-512. ResearchGate. doi: 10.1037/0021-9010.86.3.499.

Dice, T. F., Simmons, J. and Wolfenden, O. (2018) 'Responding to epidemics and pandemics: The role of human services professionals', *Journal of Human Services*, 38(1), pp. 45-59. Education Research Complete. Available at:

http://search.ebscohost.com/login.aspx?direct=true&AuthType=ip,cookie,shib&db=ehh&AN=14285 8009&site=eds-live&scope=site [Accessed: 14 March 2021].

Dollard, M. F., Dormann, C., Boyd, C. M., Winefield, H. R. and Winefield, A. H. (2003) 'Unique aspects of stress in human service work', *Australian Psychologist*, 38(2), pp. 84-91. Education Research Complete. doi: 10.1080/00050060310001707087

Dublin Region Homeless Executive (2019) *Dublin Region Homeless Executive confirms the official Spring Rough Sleeping Count*. Available at: <u>https://www.homelessdublin.ie/content/files/30th-April-RSC-Spring-2019.pdf</u> [Accessed: 06 March 2021].

Dublin Region Homeless Executive (2019) *The homelessness action plan framework for Dublin, 2019–202.* Available at: https://www.homelessdublin.ie/content/files/Homelessness-Action_Plan-2019-2021.pdf [Accessed: 5 March 2021].

Dublin Region Homeless Executive (2021) *Facts*. Available at: https://www.homelessdublin.ie/info/facts [Accessed: 3 April 2021].

Dyrbye, L. and Shanafelt, T. (2016) 'A narrative review on burnout experienced by medical students and residents', *Medical Education*, 50(1), pp. 132-149.Psychological and Behavioural Science Collection. doi: 10.1111/medu.12927.

Edelwich, J. and Brodsky, A. (1980). *Burnout: Stages of disillusionment in the helping profession*. New York: Human Sciences Press.

Edwards, J. R. (2020) 'The peaceful coexistence of ethics and quantitative research', Journal of Business Ethics, 167(1), pp. 31–40. Bussiness Source Ultimate. doi: 10.1007/s10551-019-04197-6.

Elfering, A., Grebner, S., Leitner, M., Hirschmüller, A., Kubosch, E.J. and Baur, H. (2017) 'Quantitative work demands, emotional demands, and cognitive stress symptoms in surgery nurses', *Psychology, Health and Medicine*, 22(5), pp. 604-610. Academic Search Complete. doi: 10.1080/13548506.2016.1200731

Farmer, D. (2020) 'Teacher attrition: The impacts of stress', *Delta Kappa Gamma Bulletin*, 87(1), pp. 41–50. Education Research Complete. Available at:

http://search.ebscohost.com/login.aspx?direct=true&AuthType=ip,cookie,shib&db=ehh&AN=14631 4001&site=eds-live&scope=site [Accessed: 10 March 2021].

FEANTSA (2005) *ETHOS, Typology on Homelessness and Housing Exclusion*. Available at: https://www.feantsa.org/download/ethos2484215748748239888.pdf [Accessed: 3 April 2021].

FEANTSA (2014) *Extent and profile of homelessness in European member states*. Available at: <u>https://www.feantsaresearch.org/download/feantsa-studies_04web24451152053828533981.pdf</u> [Accessed: 2 March 2021].

Ferdian, A., Luturlean, B.S., Zhafira, K.D. and Izumi N. K. (2020) 'The Impact of work stress on turnover intention in indonesia: Is there a mediation from employee' job satisfaction?', *Journal of*

Management and Marketing Review, 5(1), 31-40. Business Source Ultimate. doi:10.35609/jmmr.2020.5.1(3)

Focus Ireland (2019) *Annual report 2019*. Available at: <u>https://reports.focusireland.ie/</u> [Accessed: 6 March 2021].

Focus Ireland (2021) *Our services list*. Available at: <u>https://www.focusireland.ie/our-stories/our-services-list/</u> [Accessed: 3 April 2021].

Focus Ireland (2021) *What is homelessness?*. Available at: <u>https://www.focusireland.ie/get-</u>involved/schools/student-resources/what-is-homelessness/ [Accessed: 3 April 2021].

Freudenberger, H. J. (1974) 'Staff Burn-Out', *Journal of Social Issues*, 30(1), pp. 159-165. Wiley. doi: 10.1111/j.1540-4560.1974.tb00706.x.

Gabriel, K. P. and Aguinis, H. (2021) 'How to prevent and combat employee burnout and create healthier workplaces during crises and beyond', *Business Horizons*. Science Direct. doi: 10.1016/j.bushor.2021.02.037.

Ghanizadeh, A. and Jahedizadeh, S. (2015) 'Teacher burnout: A review of sources and ramifications', British Journal of Education, Society and Behavioural Science, 6(1), pp. 24-39. ResearchGate. doi: 10.9734/BJESBS/2015/15162.

Goldkuhl, G. (2012) 'Pragmatism vs interpretivism in qualitative information systems research', *European Journal of Information Systems*, 21(1), pp. 135-146. Semantic Scholar. doi: 10.1057/ejis.2011.54

Gonzalez-Mulé, E., Kim, M. and Ryu, J. W. (2020) 'A meta-analytic test of multiplicative and additive models of job demands, resources, and stress', *Journal of Applied Psychology*. APA PsycArticles. doi: 10.1037/apl0000840.supp (Supplemental).

Granster, D. C. and Rosen, C. C. (2013) 'Work stress and employee health a multidisciplinary review', *Journal of Management*, 39(5), pp. 1085-1122. ResearGate. doi: 10.1177/0149206313475815

Green, A., Albanese, B., Shapiro, N. and Aarons, G. (2014). The 'roles of individual and organizational factors in burnout among community-based mental health service providers', Psychological services, 11(1), pp. 41-49. ResearchGate. doi: 10.1037/a0035299.

Greene, G. (1961) A burnt-out case. London: William Heinemann

Guba, E. and Lincoln, Y. (1994) 'Competing paradigms in qualitative research' in Denzin, N. K. and Lincoln, Y. S. (eds.) *Handbook of qualitative research*, 3rd edn. Thousand Oaks: SAGE Publications, pp. 105-117.

Guest, G., Namey, E. and Mitchell, M. (2013) *Collecting qualitative data: A field manual for applied research*. Thousand Oaks: SAGE Publications.

Gupta, R.K. and Awasthy, R. (2015) Qualitative Research in Management: Methods and Experiences. New Delhi: Sage Publications Ltd. Available at:

http://search.ebscohost.com/login.aspx?direct=true&AuthType=ip,cookie,shib&db=nlebk&AN=9718 13&site=eds-live&scope=site [Accessed: 9 December 2020].

Hatch, D. J., Potter, G. G., Martus, P., Rose, U. and Freude, G. (2019) 'Lagged versus concurrent changes between burnout and depression symptoms and unique contributions from job demands and job resources', *Journal of Occupational Health Psychology*, 24(6), pp. 617-628. APA PsycArticles. doi: 10.1037/ocp0000170.supp (Supplemental).

Herbert, J. and Caldwell, T. (2015) 'Clinical Supervision', in Chan, F, Berven, N. L. and Thomas, K. R. (eds.) *Counseling Theories and Techniques for Rehabilitation and Mental Health Professionals*. 2nd edn. United States of America: Springer, pp. 443-461.

Ho, S. K. (2016) 'Relationships among humour, self-esteem, and social support to burnout in school teachers', *Social Psychology of Education*, 19(1), pp. 41-59. Gale Academic OneFile. doi: 10.1007/s11218-015-9309-7.

Hoffman, S., Palladino, J. M. and Barnett, J. (2007) 'compassion fatigue as a theoretical framework to help understand burnout among special education teachers', *Online Submission*, 2(1), pp. 15-22. ERIC. Available at:

http://search.ebscohost.com/login.aspx?direct=true&AuthType=ip,cookie,shib&db=eric&AN=ED558 015&site=eds-live&scope=site [Accessed: 20 March 2021].

Hollet-Haudebert, S., Mulki, J. P. and Fournier, C. (2011) 'Neglected burnout dimensions: Effect of depersonalization and personal nonaccomplishment on organizational commitment of salespeople', *The Journal of Personal Selling and Sales Management*, 31(4), pp. 411-428. JSTOR Journals. doi: 10.2753/PSS0885-3134310404.

Hopper, E., L. Bassuk, E. and Olivet, J. (2010) 'Shelter from the storm: Trauma-informed care in homelessness services settings', *The Open Health Services and Policy Journal*, 3(1), pp. 80-100. Suplemental Index. doi: 10.2174/1874924001003020080

Hu, N. C., Chen, J. D. and Cheng, T. J. (2014) 'The associations between long working hours, physical inactivity, and burnout', *Journal of Occupational and Environmental Medicine*, 58(5), pp. 514-518. JSTOR. doi: 10.1097/JOM.000000000000715.

Hu, Q. and Schaufeli, W. B. (2011) 'Job Insecurity and remuneration in chinese family-owned business workers', *Career Development International*, 16(1), pp. 6-19. ERIC. Available at: http://search.ebscohost.com/login.aspx?direct=true&AuthType=ip,cookie,shib&db=eric&AN=EJ9230 04&site=eds-live&scope=site [Accessed: 13 March 2021].

Hulsegge, G., van Mechelen, W., Proper, K. I., Paagman, H. and Anema, J. R. (2020) 'Shift work, and burnout and distress among 7798 blue-collar workers', *International Archives of Occupational and Environmental Health*, 93(8), pp. 955-963. ResearchGate doi: 10.1007/s00420-020-01536-3.

Hyde, K. F. (2000) 'Recognising deductive processes in qualitative research', *Qualitative Market Research: An International Journal*, 3(2), pp. 82–90. Emerald Insight. doi: 10.1108/13522750010322089.

Ireland. Housing Act 1988. No.28, Section 2. Dublin: Statutory Office

James, S. M., Honn, K. A., Gaddameedhi, S. and Van Dongen, H. P. A. (2017) 'Shift work: Disrupted circadian rhythms and sleep-implications for health and well-being', *Current sleep medicine reports*, 3(2), pp. 104-112. MEDLINE. doi: 10.1007/s40675-017-0071-6.

Javadi, M. and Zarea, K. (2016) 'Understanding thematic analysis and its pitfall', *Journal of Client Care*, 1(1), pp. 33-39. ResearGate. doi: 10.15412/J.JCC.02010107.

Jin, C. L., Chen, T., Wu, S. Y. and Yang, Y. L. (2020) 'Exploring the impact of stress on burnout: A mathematical model and empirical research', *Discrete Dynamics in Nature and Society*, 1(1), pp. 1-8. Academic Search Complete. doi: 10.1155/2020/3475324.

Kankam, P. K. (2019) 'The use of paradigms in information research', *Library and Information Science Research*, 41(2), pp. 85-92. Science Direct. doi: 10.1016/j.lisr.2019.04.003.

Kar, S. and Suar, D. (2014) 'role of burnout in the relationship between job demands and job outcomes among indian nurses', *Vikalpa: The Journal for Decision Makers*, 39(4), pp. 23-37. Business Source Ultimate. doi: 10.1177/0256090920140403.

Kheswa, J. G. (2019) 'Factors and effects of work-related stress and burnout on the well-being of social workers in the Eastern Cape province, South Africa', South African Journal of Industrial Psychology. 45(1), pp.1-10. Business Source Ultimate. doi: 10.4102/sajip.v45i0.1661

Kidd, S. A., Miner, S., Walker, D. and Davidson, L. (2007) 'Stories of working with homeless youth: On being "mind-boggling", *Children and Youth Services Review*, 29(1), pp. 16–34. ScienceDirect. doi: 10.1016/j.childyouth.2006.03.008.

Kim, B., Liu, L., Ishikawa, H. and Park, S. H. (2019) 'Relationships between social support, job autonomy, job satisfaction, and burnout among care workers in long-term care facilities in Hawaii', *Educational Gerontology*, 45(1), pp. 57-68. Psychology and Behavioral Sciences Collection. doi: 10.1080/03601277.2019.1580938.

Kim, H. J., Shin, K. H. and Umbreit, W. T. (2007) 'Hotel job burnout: The role of personality characteristics', *International Journal of Hospitality Management*, 26(2), pp. 421-434. ScienceDirect. doi: 10.1016/j.ijhm.2006.03.006.

Kristensen, T., Borritz, M., Villadsen, E. and Christensen, K. (2005) 'The Copenhagen burnout inventory: A new tool for the assessment of burnout', *Work and Stress*, 19(3), pp. 192-207. Business Source Ultimate. doi: 10.1080/02678370500297720.

Kutz, A. (2017) 'How to avoid destroying your employees and organisations due to burnouts, braindrain and fading performance? Stop double bind-communication in your organisation!', *Journal of Organization Design*, 6(1), pp. 1-12. Business Source Ultimate. doi: 10.1186/s41469-017-0015-0.

Lee, J., Lim, N., Yang, E. and Lee, S. (2011) 'Antecedents and consequences of three dimensions of burnout in psychotherapists: A meta-analysis', *Professional Psychology, Research and Practice*, 42(3), pp. 252-258. Gale Academic OneFile Select. Available at:

http://search.ebscohost.com/login.aspx?direct=true&AuthType=ip,cookie,shib&db=edsgao&AN=eds gcl.259679160&site=eds-live&scope=site [Accessed: 9 March 2021].

Lemieux, C. A. and Taylor, E. P. (2019) 'An exploratory study on the factors affecting the mental health and well-being of frontline workers in homeless services', *Health and Social Care in the Community*, 27(4), pp. 367-378. Academic Search Complete. doi: 10.1111/hsc.12738.

Lenzi, M., Santinello, M., Gaboardi, M., Disperati, F., Vieno, A., Calcagnì, A., Greenwood, R. M., Rogowska, A. M., Wolf, J. R., Loubière, S., Beijer, U., Bernad, R., Vargas-Moniz, M. J., Ornelas, J., Spinnewijn, F. and Shinn, M. (2020) 'Factors associated with providers' work engagement and burnout in homeless services: A cross-national study', *American Journal of Community Psychology*. Wiley Online Library. doi: 10.1002/ajcp.12470. Li, Y., Li, Y. and Castaño, G. (2019) 'The impact of teaching-research conflict on job burnout among university teachers: An integrated model', *International Journal of Conflict Management*, 31(1), pp. 76–90. Emerald Insight. doi: 10.1108/IJCMA-05-2019-0080.

Liu, L. (2016) 'Using generic inductive approach in qualitative educational research: A case study analysis', journal of education and learning, 5(2), pp. 129-135. ERIC. Available at: http://search.ebscohost.com/login.aspx?direct=true&AuthType=ip,cookie,shib&db=eric&AN=EJ1097 415&site=eds-live&scope=site [Accessed: 18 March 2021].

Louison Vang, M., Shevlin, M., Hansen, M., Lund, L., Askerod, D., Bramsen, R. H. and Flanagan, N. (2020) 'Secondary traumatisation, burn-out and functional impairment: Findings from a study of Danish child protection workers', *European Journal of Psychotraumatology*, 11(1), pp. 1-11. Academic Search Complete. doi: 10.1080/20008198.2020.1724416.

Maarouf, H. (2019) 'Pragmatism as a supportive paradigm for the mixed research approach: conceptualizing the ontological, epistemological, and axiological stances of pragmatism', *International Business Research*, 12(9), pp. 1-12. ResearchGate. doi: 10.5539/ibr.v12n9p1

Maidasari, R., Absah, Y. and Siahaan, E. (2018) 'An analysis of the implication of workload and social support toward burnout', *Academic Journal of Economic Studies*, 4(3), pp. 187-98. Business Source Ultimate. Available at:

http://search.ebscohost.com/login.aspx?direct=true&AuthType=ip,cookie,shib&db=bsu&AN=13184 4718&site=eds-live&scope=site [Accessed: 11 March 2021].

Malmqvist, J., Hellberg, K., Möllås, G., Rose, R. and Shevlin, M. (2019) 'Conducting the pilot study: A neglected part of the research process? methodological findings supporting the importance of piloting in qualitative research studies', *International Journal of Qualitative Methods*, 18(1), pp. 1-11. Directory of Open Access Journals. doi: 10.1177/1609406919878341.

Marashi, H. and Bani-Ardalani, M. (2017) 'Using clinical supervision to lower efl teachers' burnout', *Issues in Language Teaching*, 6(2), pp. 261-289. Directory of Open Access Journals. doi: 10.22054/ilt.2019.44172.404.

Marc, C., Dimény, J. M. and Bacter, C. (2019) 'The social worker-client relationship: Difficulties and solutions', *Bulletin of the Transilvania University of Brasov*, 12(2), pp. 377-386. Academic Search Complete. doi: 10.31926/but.ssl.2019.12.61.2.20.

Maslach, C. (1978) 'Job burnout: How people cope', *Public Welfare*, 36(1), pp. 56-58. ResearchGate. Available at: https://www.researchgate.net/publication/285102253_Job_burnout_How_people_cope [Accessed: 6 March 2021].

Maslach, C. and Jackson, S. E. (1981) 'The measurement of experienced burnout', *Journal of Occupational Behaviour*, 2(2), pp. 99-113. Wiley. doi:10.1002/job.4030020205

Maslach, C. and Leiter, M. P. (2008) 'Early predictors of job burnout and engagement', *Journal of Applied Psychology*, 93(3), pp. 498–512. APA Psyarticles. doi: 10.1037/0021-9010.93.3.498.

Maslach, C. and Leiter, M. P. (2016) 'Understanding the burnout experience: Recent research and its implications for psychiatry', *World Psychiatry: Official Journal of the World Psychiatric Association*, 15(2), pp. 103-111. ResearchGate. doi: 10.1002/wps.20311.

Maslach, C., and Leiter, M. P. (2007) 'Burnout', in Fink, G. (ed.), Handbook of stress: Vol. 1. Stress: Concepts, cognition, emotion, and behavior. Unites States of America: Elsevier Academic Press, pp. 351-357.

Maslach, C., Schaufeli, W. B. and Leiter, M. P. (2001) 'Job burnout', *Annual Review of Psychology*, 52(1), pp. 397-422. Academic Search Complete. doi: 10.1146/annurev.psych.52.1.397.

McCormack, N. and Cotter, C. (2013) *Managing burnout in the workplace: A guide for information professionals*. EBook Business Collection. Available at:

http://search.ebscohost.com/login.aspx?direct=true&AuthType=ip,cookie,shib&db=e020mww&AN= 670972&site=eds-live&scope=site [Accessed: 6 December 2020].

McFadden, P. (2020) 'Two sides of one coin? relationships build resilience or contribute to burnout in child protection social work: Shared perspectives from Leavers and Stayers in Northern Ireland', *International Social Work*, 63(2), pp. 164-176. Academic Search Complete. doi: 10.1177/0020872818788393.

McFadden, P., Mallett, J. and Leiter, M. (2018) 'Extending the two-process model of burnout in child protection workers: The role of resilience in mediating burnout via organizational factors of control, values, fairness, reward, workload, and community relationships', *Stress & Health: Journal of the International Society for the Investigation of Stress*, 34(1), pp. 72-83. Wiley. doi: 10.1002/smi.2763.

McFadden, P., Mallett, J., Campbell, A. and Taylor, B. (2019) 'Explaining self-reported resilience in child-protection social work: The role of organisational factors, demographic information and job characteristics', *British Journal of Social Work*, 49(1), pp. 198-216. ResearchGate. Available at: http://search.ebscohost.com/login.aspx?direct=true&AuthType=ip,cookie,shib&db=edb&AN=13469 1332&site=eds-live&scope=site [Accessed: 13 March 2021].

McNicholas, F., Sharma, S., O'Connor, C. and Barret E. (2020) 'Burnout in consultants in child and adolescent mental health services (CAMHS) in Ireland: A cross-sectional study', *BMJ open*, 10(1), pp. 303-354. Medline. doi: 10.1136/bmjopen-2019-030354.

Medina, A. and Beyebach, M. (2014) 'The impact of solution-focused training on professionals' beliefs, practices and burnout of child protection workers in Tenerife island', *Child Care in Practice*, 20(1), pp. 7-36. Academic Search Complete. doi: 10.1080/13575279.2013.847058.

Mehrad, A. (2019) "Importance of job burnout effects on employees' attitude and performance at workplace: A review', International Journal for Research in Educational Studies, 5(5), pp. 14-19. ResearchGate. Available at:

https://www.researchgate.net/publication/333449555 Importance of Job Burnout Effects on E mployees%27_Attitude_and_Performance_at_Workplace_A_Review [Accessed: 10 April 2021].

Mette, J., Wirth, T., Nienhaus, A., Harth, V. and Mache, S (2020) "I need to take care of myself": A qualitative study on coping strategies, support and health promotion for social workers serving refugees and homeless individuals', *Journal of Occupational Medicine and Toxicology*, 15(1), pp. 1-15. Academic Search Complete. doi: 10.1186/s12995-020-00270-3.

Moffat, C. T. (2011) 'Helping those in need: Human service workers', *Occupational Outlook Quarterly*, 55(3), pp. 22–32. ERIC. Available at:

http://search.ebscohost.com/login.aspx?direct=true&AuthType=ip,cookie,shib&db=eric&AN=EJ9459 70&site=eds-live&scope=site [Accessed: 14 March 2021].

Mullen, J. and Leginski, W. (2010) 'Building the capacity of the homeless service workforce', *The Open Health Services and Policy Journal*, 3(1), pp. 101–110. Semantic Scholar. doi:10.2174/1874924001003020101

Nagy-Borsy, E., Vagi, Z., Skerlecz, P., Szeitl, B., Kiss, I. and Rakosy, Z. (2021) 'Health status and health behaviour of the Hungarian homeless people', *Archives of Public Health*, 79(15), pp.1-12. Gale Academic OneFile. doi: 10.1186/s13690-021-00534-2.

Neff, L. A. and Karney, B. R. (2007) 'Stress crossover in newlywed marriage: A longitudinal and dyadic perspective', *Journal of Marriage and Family*, 69(3), pp. 594–607. Academia. doi: 10.1111/j.1741-3737.2007.00394.x

Nguyen C. T. and Tran T. L. T. (2015) 'The interconnection between interpretivist paradigm and qualitative methods in education', *American Journal of Educational Science*, 1(2), pp. 24-27. Semantic Scholar. Available at: <u>https://www.semanticscholar.org/paper/The-Interconnection-</u>

Between-Interpretivist-Paradigm-Thanh-Thanh/79e6888e672cf2acf8afe2ec21fd42a29b2cbd90

[Accessed: 17 March 2021].

Nisar, S. K. and Rasheed, M. I. (2020) 'Stress and performance: Investigating relationship between occupational stress, career satisfaction, and job performance of police employees', *Journal of Public Affairs*, 20(1), pp. 1-9. doi: 10.1002/pa.1986.

Novyk, L. and Mazur, T. (2021) 'Studying the impact of professional interaction on social workers' emotional burnout', Broad Research in Artificial Intelligence and Neuroscience, 12(1), pp. 172-182. Science Direct. doi: 10.18662/brain/12.1/177.

O'Brien, E. and Linehan, C. (2019) 'Problematizing the authentic self in conceptualizations of emotional dissonance', *Human Relations*, 72(9), pp. 1530-1556. Business Source Ultimate. doi: 10.1177/0018726718809166.

Onyemah, V. (2008) 'Role ambiguity, role conflict, and performance: empirical evidence of an inverted-u relationship', *Journal of Personal Selling and Sales Management*, 28(3), pp. 2993-3013. Business Source Ultimate. doi: 10.2753/PSS0885-3134280306.

Panari, C., Caricati, L., Pelosi, A. and Rossi, C. (2019) 'Emotional exhaustion among healthcare professionals: The effects of role ambiguity, work engagement and professional commitment', *Acta bio-medica: Atenei Parmensis*, 90(6), pp. 60–67. MEDLINE. doi: 10.23750/abm.v90i6-S.8481.

Pearsall, M. J., Ellis, A. P. J. and Stein, J. H. (2009) 'Coping with challenge and hindrance stressors in teams: Behavioral, cognitive, and affective outcomes', *Organizational Behavior and Human Decision Processes*, 109(1), pp. 18-28. ScienceDirect. doi: 10.1016/j.obhdp.2009.02.002.

Perlman, B. and Hartman, E. A. (1982). Burnout: Summary and future research. *Human Relations*, 35(4), pp. 283-305. Sage Journals. doi: 10.1177/001872678203500402

Pines, A. and Kafry, D. (1978) 'Occupational tedium in the social services'. *Social Work*, 23(6), pp. 499-507. Psychology and Behavioral Sciences Collection. Available at:

http://search.ebscohost.com/login.aspx?direct=true&AuthType=ip,cookie,shib&db=pbh&AN=52750 34&site=eds-live&scope=site [Accessed: 8 March 2021].

Podsakoff, N. P., LePine, J. A. and LePine, M. A. (2007) 'Differential challenge stressor-hindrance stressor relationships with job attitudes, turnover intentions, turnover, and withdrawal behavior: A meta-analysis', *Journal of Applied Psychology*, 92(2), pp. 438-454. APA PsycArticles. doi: 10.1037/0021-9010.92.2.438.

Price, J. H. and Murnan, J. (2004) 'Research limitations and the necessity of reporting them', *American Journal of Health Education*, 35(2), pp. 66-67. Taylor and Francis Online. doi: 10.1080/19325037.2004.10603611.

Qu, S. Q. and Dumay, J. (2011) 'The qualitative research interview', Qualitative Research in Accounting & Management, 8(3), pp. 238–264. Emeral Insight. doi: 10.1108/11766091111162070.

Queirós, A., Faria, D. and Almeida, F. (2017) 'Strengths and limitation of qualitative and quantitative research methods', European Journal of Education Studies, 3(9), pp. 369-387. Research Gate. doi: 10.5281/zenodo.887089.

Quimby, E. (2012) Doing Qualitative Community Research: Lessons for Faculty, Students and Communities. Sharjah: Bentham Science Publishers. Available at:

http://search.ebscohost.com/login.aspx?direct=true&AuthType=ip,cookie,shib&db=nlebk&AN=5007 04&site=eds-live&scope=site [Accessed: 9 December 2020].

Ramonienė, L. and Gorbatenko, O. (2013) 'Managing occupational stress in human service nonprofit organizations in mykolaiv, Ukraine', *Organizations and Markets in Emerging Economies*, 4(2), pp. 64-96. Business Source Ultimate. doi: 10.15388/omee.2013.4.2.14250.

Rasheed-Karim, W. (2020) 'The effect of stressful factors, locus of control and age on emotional labour and burnout among further and adult education teachers in the U.K', *International Journal of Emerging Technologies in Learning*, 15(24), pp. 26-37. Education Research Complete. doi: 10.3991/ijet.v15i24.19305.

Robbins, S.P. and Judge, T. (2017) Organizational behaviour. 15th edn. London: Pearson

Robelski, S. Sette, J., Wirth, T., Kiepe, N., Nienhaus, A., Harth, V. and Mache, S. (2020) '(Un)bounded social work? analysis of working conditions in refugee and homeless aid in relation to perceived job stress and job satisfaction', *International Journal of Environmental Research and Public Health*, 17(2), pp. 1-20. MEDLINE. doi: 10.3390/ijerph17020601.

Ruch, G. (2007) 'Reflective practice in contemporary child-care social work: The role of containment', *The British Journal of Social Work*, 37(4), pp. 659–680. JSTOR Journals. Available at: <u>https://www.jstor.org/stable/23722588?seg=1</u> [Accessed: 13 March 2021].

Rudman, A., Arborelius, L., Dahlgren, A., Finnes, A. and Gustavsson, P. (2020) 'Consequences of early career nurse burnout: A prospective long-term follow-up on cognitive functions, depressive symptoms, and insomnia', *EClinicalMedicine*, 27(1), pp. 2-8. ScienceDirect. doi: 10.1016/j.eclinm.2020.100565.

82

Salami, S. O. and Ajitoni, S. O. (2016) 'Job characteristics and burnout: The moderating roles of emotional intelligence, motivation and pay among bank employees', *International Journal of Psychology*, 51(5), pp. 375-382. Business Source Ultimate. doi: 10.1002/ijop.12180.

Salvagioni, D. A. J., Melanda, F. N., Mesas, A. E., González, A. D., Gabani, F. L. and Andrade, S. M. (2017) 'Physical, psychological and occupational consequences of job burnout: A systematic review of prospective studies', *Plos One*, 12(10), pp. 1-29. Academic Search Complete. doi: 10.1371/journal.pone.0185781.

Satish, P. and Pestonjee, D. M. (2013) *Stress and work: Perspectives on understanding and managing stress*. Sage Publications. Available at:

http://search.ebscohost.com/login.aspx?direct=true&AuthType=ip,cookie,shib&db=nlebk&AN=1214 390&site=eds-live&scope=site [Accessed: 10 March 2021].

Saunders, M., Lewis, P. and Thornhill, A. (2019) *Research methods for business students*. 8th edn. VLeBooks. Available at:

http://search.ebscohost.com/login.aspx?direct=true&AuthType=ip,cookie,shib&db=cat05743a&AN= nci.33365&site=eds-live&scope=site [Accessed: 17 March 2021].

Schaufeli, W. (2017) 'Applying the job demands-resources model: A 'how to' guide to measuring and tackling work engagement and burnout', *Organizational Dynamics*, 46, pp. 120-132. Science Direct. doi: 10.1016/j.orgdyn.2017.04.008

Schaufeli, W. and Taris, T. (2014) 'A critical review of the job demands-resources model: Implications for improving work and health' *Bridging Occupational, Organizational and Public Health: A Transdisciplinary Approach*, 1(1), pp. 43-68, Semantic Scholar. doi: 10.1007/978-94-007-5640-3_4

Schaufeli, W.B., Salanova, M., González-Romá, V. and Bakker, A. B. (2002) 'The measurement of engagement and burnout: A two sample confirmatory factor analytic approach', *Journal of Happiness Studies*, 3(1), pp. 71-92. Psychology and Behavioral Sciences Collection. doi: 10.1023/A:1015630930326.

Seljemo, C., Viksveen, P. and Ree, E. (2020) 'The role of transformational leadership, job demands and job resources for patient safety culture in Norwegian nursing homes: A cross-sectional study', *BMC Health Services Research*, 20(1), pp. 1-8. Scopus. doi: 10.1186/s12913-020-05671-y.

Sharma, S. and Singh, J. (2016) 'Identification of sources of job stress: A study of bank marketing executives', *Journal of Organisation and Human Behaviour*, 5(2), pp. 23-28. Business Source Ultimate. doi: 10.21863/johb/2016.5.2.033.

Shirom, A. (2009) 'Burnout and health: Expanding our knowledge', *Stress & Health: Journal of the International Society for the Investigation of Stress*, 25(4), pp. 281–285.Wiley. doi: 10.1002/smi.1283.

Simon Communities of Ireland (2021) *About homelessness*. Available at: https://www.simon.ie/Homelessness/Homelessness.aspx [Accessed: 3 April 2021].

Singh, R. K. and Jain, M. (2017) 'Efficacy of self-management techniques in reducing perceived occupational stress among nursing staff', *Indian Journal of Positive Psychology*, 8(3), pp. 360–365. Academic Search Complete. Available at:

http://search.ebscohost.com/login.aspx?direct=true&AuthType=ip,cookie,shib&db=a9h&AN=12608 0984&site=eds-live&scope=site [Accessed: 10 March 2021].

Stelmokienė, A., Genevičiūtė-Janonė, G., Gustainienė, L. and Kovalčikienė, K. (2019) 'Job demandsresources and personal resources as risk and safety factors for the professional burnout among university teachers', *Pedagogy Studies / Pedagogika*, 134(2), pp. 25–44. doi: 10.15823/p.2019.134.2.

Sulaiman, C. F. C., Henn, P., Smith, S. and O'Tuathaigh, C. M. P. (2017) 'Burnout syndrome among non-consultant hospital doctors in Ireland: relationship with self-reported patient care', *International Journal for Quality in Health Care: Journal of the International Society for Quality in Health Care*, 29(5), pp. 679-684. Medline. doi: 10.1093/intqhc/mzx087.

Sumner, R. C. and Kinsella, E. L. (2021) 'Grace under pressure: Resilience, burnout, and wellbeing in frontline workers in the United Kingdom and Republic of Ireland during the sars-cov-2 pandemic', *Frontiers in Psychology*, 11(1), pp.1-18. Gale Academic OneFile Select doi: 10.3389/fpsyg.2020.576229.

Suñer-Soler, R., Grau-Martín, A., Flichtentrei, D., Prats, M., Braga., F, Font-Mayolas, S. and Gras, M. E. (2014) 'The consequences of burnout syndrome among healthcare professionals in Spain and spanish speaking latin american countries', *Burnout Research*, 1(2), pp. 82-89. Science Direct. doi: 10.1016/j.burn.2014.07.004.

Tafvelin, S., Nielsen, K., von Thiele Schwarz, U. and Stenling, A. (2019) 'Leading well is a matter of resources: Leader vigour and peer support augments the relationship between transformational leadership and burnout', *Work and Stress*, 33(2), pp. 156-172. Psychology and Behavioral Sciences Collection. doi: 10.1080/02678373.2018.1513961.

Taherdoost, H. (2016) 'Sampling methods in research methodology; how to choose a sampling technique for research', *International Journal of Academic Research in Management*, 5(2), pp. 18-27. ResearchGate. doi: 10.2139/ssrn.3205035.

The World Health Organisation (2007) *Protecting workers' health series no. 3, work organization and stress*. Available at: <u>https://www.who.int/occupational_health/publications/stress/en/</u> [Accessed: 10 March 2021].

Thomas, D. R. (2006) 'A general inductive approach for analysing qualitative evaluation data', *American Journal of Evaluation*, 27(2), pp. 237-246. ERIC. Available at: http://search.ebscohost.com/login.aspx?direct=true&AuthType=ip,cookie,shib&db=eric&AN=EJ7366 30&site=eds-live&scope=site [Accessed: 18 March 2021].

Thomas, M., Kohli, V. and Choi, J. (2014) 'Correlates of job burnout among human services workers: Implications for workforce retention', *Journal of Sociology & Social Welfare*, 4(1), pp. 69-89. EBScohost. Available at:

<u>https://scholarworks.wmich.edu/cgi/viewcontent.cgi?article=3983&context=jssw&httpsredir=1&ref</u> <u>erer</u>= [Accessed: 5 March 2021].

Totawar, A. K. and Nambudiri, R. (2012) 'An overview of depersonalization in the organizational context'. *Indore Management Journal*, 4(2), pp. 64-72. SCRIBD. Available at: https://www.scribd.com/document/249407234/Vol-4-Issue-2-Full [Accessed: 9 March 2021].

Tunc, T. and Kutanis, R.O. (2009) 'Role conflict, role ambiguity, and burnout in nurses and physicians at a university hospital in Turkey', *Nursing and Health Sciences*, 11 (4), pp. 410-416. Semantic Scholar. doi: 10.1111/j.1442-2018.2009.00475.x

Turan, I. (2020) 'Thematic vs chronological history teaching debate: A social media research', *Journal of Education and Learning*, 9(1), pp. 205-216. ERIC. Available at:

http://search.ebscohost.com/login.aspx?direct=true&AuthType=ip,cookie,shib&db=eric&AN=EJ1241 024&site=eds-live&scope=site [Accessed: 21 March 2021].

Turek, D. (2020) 'When does job burnout not hurt employee behaviours?', *Journal of Organizational Effectiveness: People and Performance,* 8(1), pp. 59-79. Emerald Insight. doi: 10.1108/JOEPP-04-2020-0055.

Valcour, M. (2016) 'Beating burnout', *Harvard Business Review*, 94(11), pp. 98-101. Business Source Ultimate. Available at:

http://search.ebscohost.com/login.aspx?direct=true&AuthType=ip,cookie,shib&db=bsu&AN=11886 6543&site=eds-live&scope=site [Accessed: 9 March 2021].

Van den Tooren, M. and De Jong, J. (2014) 'Job demands-resources and employee health and wellbeing: The moderating role of contract type', Career Development International. 19(1), pp. 102-122. ResearchGate. doi: 10.1108/CDI-05-2013-0058.

Van Veldhoven, M., Van den Broeck, A., Daniels, K., Bakker, A. B., Tavares, S. M. and Ogbonnaya, C. (2020) 'Challenging the universality of job resources: Why, when, and for whom are they beneficial?', *Applied Psychology: An International Review*, 69(1), pp. 5-29. doi: 10.1111/apps.12211.

Waegemakers Schiff, J. and Lane, A. M. (2019) 'PTSD symptoms, vicarious traumatization, and burnout in front line workers in the homeless sector', *Community Mental Health Journal*, 55(3), pp. 454-462. Gale Academic OneFile. doi: 10.1007/s10597-018-00364-7.

Wallbank, S. and Hatton, S. (2011) 'Reducing burnout and stress: The effectiveness of clinical supervision', *Community Practitioner*, 84(7), pp. 21-25. ResearchGate. Available at: https://www.researchgate.net/publication/51669025_Reducing_burnout_and_stress_the_effective ness_of_clinical_supervision [Accessed: 13 March 2021].

Wallbank, S. and Robertson, N. (2008) 'Midwife and nurse responses to miscarriage, stillbirth and neonatal death: A critical review of qualitative research', *International Journal of nursing studies*, 6(3), pp. 1090-1097. ResearchGate. Available at:

https://www.researchgate.net/publication/290327232_Midwife_and_nurse_responses_to_miscarri age_stillbirth_and_neonatal_death_A_critical_review_of_qualitative_research [Accessed: 13 March 2021].

Walle, A. H. (2015) *Qualitative research in business: A practical overview*. Newcastle upon Tyne, United Kingdom: Cambridge Scholars Publishing. Available at:

http://search.ebscohost.com/login.aspx?direct=true&AuthType=ip,cookie,shib&db=nlebk&AN=1014 736&site=eds-live&scope=site [Accessed: 20 March 2021].

Wang, Q. and Wang, C. (2020) 'Reducing turnover intention: Perceived organizational support for frontline employees', *Frontiers of Business Research in China*, 14(1), pp. 1-16. Springer Open. doi: 10.1186/s11782-020-00074-6.

Williams, E. S., Rathert, C. and Buttigieg, S. C. (2020) 'The personal and professional consequences of physician burnout: A systematic review of the literature', *Medical care Research and Review*, 77(5), pp. 371-386. MEDLINE. doi: 10.1177/1077558719856787.

Wirth, T., Mette, J., Nienhaus, A., Schillmöller, Z., Harth, V. and Mache, S. (2019) "This isn't just about things, it's about people and their future": A qualitative analysis of the working conditions and strains of social workers in refugee and homeless aid', *International Journal of Environmental Research and Public Health*, 16(20), pp. 1-20. MEDLINE. doi: 10.3390/ijerph16203858.

Wirth, T., Mette, J., Prill, J., Harth, V. and Nienhaus, A. (2019) 'Working conditions, mental health and coping of staff in social work with refugees and homeless individuals: A scoping review', *Health and Social Care in the Community*, 27(4), pp. 257-269. Academic Search Complete. doi: 10.1111/hsc.12730.

World Health Organsiation (2007) Protecting Workers' Health Series No. 6, Raising awareness of stress at work in developing countries. Available at:

https://www.who.int/occupational_health/publications/pwh6/en/ [Accessed: 10 April 2021].

Wu, D. (2020) 'Relationship between job burnout and mental health of teachers under work stress', Revista Argentina de Clínica Psicológica, 29(1), pp. 310-15. Academic Search Complete. doi: 10.24205/03276716.2020.41.

Xanthopoulou, D., Bakker, A. B., Dollard, M. F., Demerouti, E., Schaufeli, W. B., Taris, T. W. and Schreurs (2007) 'When do job demands particularly predict burnout?: The moderating role of job resources', *Journal of Managerial Psychology*, 22(8), pp. 766-786. doi: 10.1108/02683940710837714.

Xu, X. and Payne, S. C. (2020) 'When do job resources buffer the effect of job demands?', *International Journal of Stress Management*, 27(3), pp. 226–240. APA PsycArticles. doi: 10.1037/str0000146.

Yang, Y. and Hayes, J. A. (2020) 'Causes and consequences of burnout among mental health professionals: A practice-oriented review of recent empirical literature', *Psychotherapy*, 57(3), pp. 426-436. APA PsycArticles. doi: 10.1037/pst0000317.

Yildirim, F. and Sait Dinc, M. (2019) 'Factors influencing burnout of the principals: A pilot study in flemish schools of Belgium', *Ekonomska Istraživanja*, 32(1), pp. 3538-3553. Directory of Open Access Journals. doi: 10.1080/1331677X.2019.1660200

Zeng, X., Zhang, X., Chen, M., Liu, J. and Wu, C. (2020) 'The influence of perceived organizational support on police job burnout: A moderated mediation model', *Frontiers in Psychology*, 1(11), pp.1-11. Directory of Open Access Journals. doi: 10.3389/fpsyg.2020.00948

Appendices

Appendix A- Interview Questions and Questions Rationale

Introduction

Hello, and thank you very much for participating in this interview. I really appreciate it. I am going to start this interview by reintroducing myself and explaining what to expect of this process. My name is Marta Franchy, and I am conducting this interview as part of my Master's in human resource management under the National College of Ireland. What I am trying to accomplish with this dissertation is to understand the feelings of burnout in the homeless services through the experiences of the employees, as this field hasn't really being explored. During this interview, I am going to ask you three sets of questions. The first part of the interview will be focused on your background; Then we will pass to the second part in which we will discuss how are you feeling emotionally towards your job, and for the third and final part, I will be asking you more specific questions about the work you carry out and your organisation. The interview will last for about 45 minutes to 1 hour. If at some point you decide you don't want to continue with the interview, please voice your concerns, and we will stop immediately. I wanted to highlight again, as mentioned before, that this interview is anonymous, and no personal information o recognisable characteristics will be share in the research. Moreover, as part of the process, you will be provided with a complete transcription of this interview, and if you wish, I will make a copy of the final thesis available for you. Do you have any questions?

Ok, so let's start ...

Demographics

- Age?
- Gender?
- Education?
- What is your job title, and what does your job entail?
- For how long have you been working in the organisation?
- Have you work in any other homeless services? For how long?
- What is your motivations to work in this field?

Burnout's dimensions

- a. What do you understand for burnout?
- b. Emotional exhaustion
 - Do you feel burnout? How?
 - Do you feel exhausted after finishing your job for the day?
 - Are your breaks like annual leave, enough to feel energize to go back to work?
 - How is your work life balance? Do you bring your work problems home?
- c. Cynicism
 - Do you ever treat your customers in a depersonalized way, line numbers or cases?
 - Do you think you are engaged with the organisation?
 - Have you ever taken a sick day without being sick have a break?
 - Do you ever take longer breaks at work than you should?
- d. Ineficaccy
 - Do you think you are good at your job?
 - Do you think you are valuable for your team/organisation?

<u>Stress</u>

- Do you feel stressed at work? Is your job stressful? Is this a normal quality of your job?
- How does the stress affect you in the work environment?
- How does the stress affect you in your personal life

Job demands

- a. Workload
 - How would you define your workload?
 - Do think you have enough time to complete all your tasks?
- b. Role conflict and role ambiguity
 - What do you think about your role? is it clearly defined?
 - Do you ever feel that you can't fulfil the expectation of clients/management? Explain how.
 - Do you have all the information necessary to carry out your job?

- a. Emotional demands
 - How would you define working with the homeless?
 - Do you ever find difficult to focus at work?
 - Does your job require an emotional effort for you to carry it out?
- b. Working hours
 - What do you think about your working hours?
 - Do you get enough rest?
 - Do you think your working hours affect your personal life?
 - Do you think your working hours affect your wellbeing?

Job Resources

- a. Peer support
 - How would you describe your relationship with your colleagues?
 - Do you feel supported by them?
 - How does this affect your work and yourself?
- b. Organisational support
 - Do you feel your organisation supports you and looks after your wellbeing?
 - Do you think the processes in your organisation are fair?
 - Do you feel your manager supports you?
 - Do you think you have good working conditions?
- c. Supervision
 - How would you define your relationship with your supervisor?
 - Do you feel you can openly share your thoughts in supervision?
 - Can you rate your supervision from 0 to 10?
 - Would you change anything about your supervision?
- d. Pay
 - Do you feel that your work is fairly remunerated?
 - Do you think the paying scheme of your organisation is fair?

To finalize:

Thank you so much for giving us your time today, your participation is greatly appreciated. Before we finish the interview I would like to ask you if iss there any question you would like to ask?

Additionally, is there anything else that you would like to talk about or to mention regarding your employment in homeless services that you think might be useful for this research?

Thank you very much for your time,

Questions Rationale

| Ref | Question | Relates back to which Research | Source of Question | Additional Comments |
|----------|---|--|---|---|
| | | Question? | | comments |
| | Theme 1: Burnout | | | |
| Q1- 2 | What do you understand for burnout? Do you feel burnout? How? | How are the providers of homeless services experiencing burnout? | There is not a universal perception of what burnout is, as elucidated in the literature review, but multiple conceptualizations by a diversity of researcher, such as Freudenberger (1974), Maslach and Jackson (1981) or Perlman and Hartman (1982). Therefore it would be of interest to investigate what the concept of burnout means for each participant. | These questions supports the understanding of what burnout means for each one of the interviewees, and the effects it effect it produces within themselves |
| | Theme 2: Emotional Exhaustion | | 1 | |
| Q3 | Do you feel exhausted after finishing your job for the day? | How are the providers of homeless services experiencing burnout? | Maslach and Leiter (2007) referred to emotional exhaustion as a dimension of burnout in which the person experience feelings of being overstretched and drained of the person's physical and emotional resources. | This question supports the identification of emotional exhaustion as one of the components of burnout being experienced by homeless services providers. |
| Q4 | Are your breaks like annual leave, enough to feel energize to go back to work? | How are the providers of homeless services experiencing burnout? | Maslach and Leiter (2007) identified that the individuals experiencing emotional exhaustion are unable to refill the energy used for their work development | This questions supports the understanding of the concept of energy replenishment and its incidence in the emotional exhaustion of the employees. |

| Ref | Question | Relates back to which Research Question? | Source of Question | Additional Comments |
|----------|---|--|---|---|
| | Theme 1: Burnout | | | |
| Q1- 2 | What do you understand for burnout? Do you feel burnout? How? | How are the providers of homeless services experiencing burnout? | There is not a universal perception of what burnout is, as elucidated in the literature review, but multiple conceptualizations by a diversity of researcher, such as Freudenberger (1974), Maslach and Jackson (1981) or Perlman and Hartman (1982). Therefore it would be of interest to investigate what the concept of burnout means for each participant. | These questions supports the understanding of what burnout means for each one of the interviewees, and the effects it effect it produces within themselves |
| | Theme 2: Emotional Exhaustion | | | |
| Q3 | Do you feel exhausted after finishing your job for the day? | How are the providers of homeless services experiencing burnout? | Maslach and Leiter (2007) referred to emotional exhaustion as a dimension of burnout in which the person experience feelings of being overstretched and drained of the person's physical and emotional resources. | This question supports the identification of emotional exhaustion as one of the components of burnout being experienced by homeless services providers. |
| Q4 | Are your breaks like annual leave, enough to feel energize to go back to work? | How are the providers of homeless services experiencing burnout? | Maslach and Leiter (2007) identified that the individuals experiencing emotional exhaustion are unable to refill the energy used for their work development | This questions supports the understanding of the concept of energy replenishment and its incidence in the emotional exhaustion of the employees. |

| Ref | Question | Relates back to which Research Question? | Source of Question | Additional Comments |
|-----|--|--|---|---|
| | Theme 2: Emotional Exhaustion | | | |
| Q5 | How is your work life balance? Do you bring your work problems home? | How are the providers of homeless services experiencing burnout? | Emotional exhaustion has been strongly linked with work- place environments with high levels of stress and the service providers' over- involvement in their work (Lee et al, 2011). | This question supports the understanding of the level of stress and involvement of the employees with their job; and the impact of these components in the capacity to replenish their energy in their time off work. |
| | Theme 3: Cynicism | - | - | |
| Q6 | Do you ever treat your customers in a depersonalized way, line numbers or cases? | How are the providers of homeless services experiencing burnout? | The second dimension of burnout, cynicism corresponds to the relational component of burnout, which produces hostile feelings and extreme detachment towards a person's employment, leading to a loss of idealism and others' dehumanisation (Maslach and Leiter, 2016; Maslach et al, 2001). | This question supports the identification of the dehumanisation component within the cynicism dimension |

| Ref | Question | Relates back to which Research Question? | Source of Question | Additional Comments | | |
|------|---|--|--|---|--|--|
| | Theme 3: Cynicism | | | | | |
| Q7 | Do you think you are engaged with the organization? | How are the providers of homeless services experiencing burnout? | The second dimension of burnout, cynicism corresponds to the relational component of burnout, which produces hostile feelings and extreme detachment towards a person's employment (Maslach and Leiter, 2016; Maslach et al, 2001). | This question supports the identification of the cynicism's component of organisational detachment | | |
| Q8-9 | Have you ever taken a sick day without being sick have a break? Do you ever take longer breaks at work than you should? | How are the providers of homeless services experiencing burnout? | Visible signs of cynicism encompass the use of pejorative language, the overuse of jargon, detachment from one's duties through longer breaks, and the over- rationalisation of situations. (Totawar and Nambudiri, 2012) | These questions supports the identification of availing of longer breaks as a component of cynicism within the group researched | | |

| Ref | Question | Relates back to which Research Question? | Source of Question | Additional Comments |
|------------|--|---|---|---|
| | Theme 3: Inefficacy | 1 | 1 | |
| Q10- 11 | Do you think you are good at your job? Do you think you are valuable for your team/organisation? | How are the providers of homeless services experiencing burnout? | The last dimension of burnout, also referred to as reduced personal accomplishment, corresponds with the self-assessment component of burnout, in which the person develops a tendency to de-valuate himself and his capacities, generating feelings of inadequacy (Maslach and Leiter, 2016). Individuals experiencing this component of burnout feel that they are incapable of succeeding in developing their job, generating strong sentiments of incompetency | These questions support the identification of the burnout's dimension of inefficacy |
| | Theme 4: Stress | | | |
| Q12- 14 | Do you feel stressed at work? Is your job stressful? Is this a normal quality of your job? How does the stress affect you in the work environment? How does the stress affect you in your personal life? | How do employees perceiving their work in terms of stress and its impact? | Jin <i>et al</i> (2020) recognized stress in their research as the main component producing burnout, while Nisar and Rasheed (2020) also identified stress as the main component of burnout resulting from highly demanding workplaces. stress sphere of the syndrome. Additionally, stress has been identified as the main component producing the burnout component of Emotional Exhaustion (Alsharif, 2020). | These set of questions support the identification of stress level in the workplace. As well as its impact in their work development and their personal life. |

| Ref | Question | Relates back to which Research Question? | Source of Question | Additional Comments |
|------------|--|---|---|--|
| | Theme 5: Job Demands | | | |
| Q15- 16 | How would you define your workload? Do think you have enough time to complete all your tasks? | Is the burnout experienced influenced by job demands and job resources? | Previous research in a healthcare setting has identified that working with clients with high needs require a significant workload, which leads to increased job stress and possible burnout (Kim et al, 2019). Equally, high workloads and time constrictions have been regarded as recurrent issue in different researches carried out in service providers of homeless services (Mette et al, 2020; Robelski et al, 2020). | These questions support the understanding of workloads in homeless services and its impact in the generation of burnout. |
| Q17- 19 | What do you think about your role? is it clearly defined? Do you have all the information necessary to carry out your job? Do you ever feel that you can't fulfil the expectation of clients/management? Explain how. | Is the burnout experienced influenced by job demands and job resources? | Role ambiguity has been defined as the perception that individuals do not have all the information necessary to carry out their tasks or work. At the same time, role conflict has also been demarcated as the feeling of unable to fulfil incompatible job demands and expectations, incapable to satisfy all parts involved in the work development (Celik, 2013; Onyemah, 2008). The non-clearness of a person's role and the expectations attached to it, could propitiate burnout due to the extra emotional investments (Panari et al, 2019). | These questions support the understanding of the clarity and conflict present in the roles carried out in homeless services and their impact in the generation of burnout. |

| Ref | Question | Relates back to which Research Question? | Source of Question | Additional Comments |
|----------------|---|---|---|--|
| | Theme 5: Job Demands | | | |
| Q20- 21 | How would you define working with the homeless? Does your job require an emotional effort for you to carry out your work? | Is the burnout experienced influenced by job demands and job resources? | Emotional demands have been defined as the work requirements that involve a constant sensitive effort during communication with customers (Stelmokienė, 2019). Emotionally demanding environments have been the cause of cognitive stress, with symptoms such as difficulty focusing or making decisions, memory retention, and thinking correctly (Elfering et al, 2017). Furthermore, previous research proposes that burnout is a consequence of prolonged exposure to emotional demands in the workplace (Xanthopoulou et al, 2007). | These questions support the understanding of the emotional effort that the providers of homeless services incur in their work development and its impact in the generation of burnout. |
| Q 22- 25 | What do you think about your working hours? Do you get enough rest? Do you think your working hours affect your personal life? Do you think your working hours affect your working hours affect your wellbeing? | Is the burnout experienced influenced by job demands and job resources? | Working hours are the hours in which the development of one's work takes place. In the literature available shift work is the source of well- being concerns as it has been linked to an overall decrease of the person's well-being in the form of physical ailments and mental health problems (Hulsegge et al, 2020; James et al, 2017). Furthermore, previous research has also found that there is a correlation between long working hours and burnout, especially in professionals under fifty years of age (Hu, Chen and Cheng, 2014). | Providers of homeless services in residential settings work hours are provided through rotas, in which some long and unsocial hours are included. These questions support the understanding of the effect that working hours have in the wellbeing of homeless service providers and its impact in the generation of burnout. |

| Ref | Question | Relates back to which Research Question? | Source of Question | Additional Comments | | |
|------------|---|---|--|--|--|--|
| | Theme 6: Job | | | | | |
| Q26- | Resources How would | Is the burnout | Peer support has been | These questions | | |
| 28 | you describe your relationship with your colleagues? Do you feel supported by them? How does this affect your work and yourself? | experienced influenced by job demands and job resources? | defined as the extent to which employees count on their colleagues to assist and support them during their work development (Tafvelin, 2019; Caringi et al, 2017). Positive relations among colleague's act as a sheltering system, protecting the employee from stress and improving their wellbeing and creating resilience. Conversely, the lack of social support in the workplace can lead to burnout, as employees do not benefit from healthy work environments where to share their work experiences (Ho, 2016). | support the understanding of the effect that the support of peers have in the wellbeing of homeless service providers and its impact in the generation of burnout. | | |
| Q29- 31 | Do you feel your organisation supports you and looks after your wellbeing? Do you think the processes in your organisation are fair? Do you feel your manager supports you? | Is the burnout experienced influenced by job demands and job reurces? | Organisational support relates to the individual's perception of how the organisation values their contributions and safeguard their welfare (Turek, 2020). If individuals don't perceive support from their companies, they acquire passive work attitudes, reduce their productivity, and may exhibit burnout symptoms (Wang and Wang, 2020). The organisational support could be shown in different working scenarios, such as the perceived justice and fairness of the organisational processes, the support provided by managers, the presence of inducements, and the existence of good working conditions (Zeng et al, 2020). | These questions support the understanding of the effect that organisational support has in the wellbeing of homeless service providers and its impact in the generation of burnout. | | |

| Ref | Question | Relates back to which Research Question? | Source of Question | Additional Comments |
|------------|--|---|--|--|
| | Theme 6: Job Resources | | | |
| Q32- 35 | How would you define your relationship with your supervisor? Do you feel you can openly share your thoughts in supervision? Can you rate your supervision from 0 to 10? Would you change anything about your supervision? | Is the burnout experienced influenced by job demands and job resources? | Supervision is the formal process of supporting, guiding, advising, and promoting growth in the employees; as well as a tool to encourage quality and wellbeing in the workplace (Marashi and Bani-Ardalani, 2017). In the provision of human services, the establishment of supervision provides for a safe space to communicate complex characteristics of the work, which are inappropriate to share with other people, protecting the employee from stress, burnout and trauma (Wallbank and Hatton, 2011). | These questions support the understanding of the effect that supervision has in the wellbeing of homeless service providers and its impact in the generation of burnout. |
| Q36- 37 | Do you feel that your work is fairly remunerated? Do you think the paying scheme of your organisation is fair? | Is the burnout experienced influenced by job demands and job resources? | Remuneration is a powerful job resource that provides motivation in the workplace and diminishes the work's unfavourable characteristics and the influence of burnout; well-paid personnel are more likely to provide extra effort, have a better attitude towards their job, and support the development of valuable personal characteristics such as good self-esteem and more confidence in carrying out their duties (Salami and Ajitoni, 2016). | These questions support the understanding of the effect that pay has in the morale of homeless service providers and its impact in the generation of burnout. |

Appendix B - Participation Request Email

Date: 16/03/2021

Dear employees of Focus Ireland,

My name is Marta Franchy, and I am completing a dissertation for the fulfilment of a MA in Human Resource Management with the National College of Ireland.

The topic chosen for my dissertation is "burnout", understood as emotional exhaustion caused by stress; and I would like to discuss the topic with employees of this organisation who may be experiencing this condition. The provision of homeless services is a challenging field to work in, and little research has been carried out within the subject chosen. Therefore, this research has been designed to know the actual opinions and experiences of providers of homeless services to hopefully bring changes to the field to improve its employees' wellbeing.

I would like to carry out interviews to explore what are the experiences of individuals undergoing this condition and what are the causes of its incidence. Interviews should take around 45 minutes to an hour and be conducted on zoom only through audio. The interviews will be carried out individually, and to provide a comfortable environment for the interviewees the laptop's camera is not required to be turned on. I would also request permission to record the interview's audio to facilitate its transcription; nevertheless, it will be deleted afterwards.

All the information provided through the interviews will be anonymised, and no identifiable information will be recorded within the dissertation. Additionally, I would like to mention that even though Focus Ireland has facilitated me to carry out my research within the organisation, the research carried out is entirely independent.

I have attached to this email an information sheet regarding the dissertation in case you would like to have more information.

I would be delighted if you would agree to contribute to this research.

If you wish to participate, you can contact me by sending me an email to the following email address: x19126158@student.ncirl.ie, or you can contact me in the following phone number 0838425508.

Thank you for your time and consideration.

Regards,

Marta Franchy

Appendix C - Participant Information Sheet

Burnout: an exploration of burnout experiences by providers of homeless services in Ireland

I would like to invite you to take part in my research study. Before you decide if you want to participate, I will provide you with some information regarding myself, this research and what your involvement would entail.

WHO I AM AND WHAT THIS STUDY IS ABOUT

My name is Marta Franchy, and I am carrying out this dissertation for the fulfilment of a MA in Human Resource Management with the National College of Ireland. The research aims to analyse the concept of burnout and its antecedents through the experiences of providers of homeless services. This dissertation was created to gain insight into homeless services providers' wellbeing due to the complex working conditions they are exposed to in their daily practice

WHAT WILL TAKING PART INVOLVE?

Participating in this research is strictly voluntary. Your participation will include an interview of 45 minutes to an hour that will take place through zoom. You may decline to answer any question you don't wish to reply to. Additionally, If, at any point, you don't desire to continue with the research, you are welcome to withdraw. With your permission, the audio of the interview will be recorded to facilitate its transcription and a copy of this will be available to you.

WILL TAKING PART BE CONFIDENTIAL?

The information provided for this research will be confidential, and no personal or identifiable information will be recorded in the dissertation. The transcription of the interviews will be stored in my personal computer, unidentifiable and password protected. If during the interview process information is disclosed that is likely to cause serious harm to others, it will have to be disclosed to the pertinent authority.

WHY HAVE YOU BEEN INVITED TO TAKE PART?

The homeless sector is a challenging field of employment due to the complex needs of the population it serves. Homeless individuals present complex needs related to experiences of marginalization, addiction, violence, among many others. Being chronically exposed to traumatic circumstances like the ones aforementioned can produce a high level of stress and exhaustion in the service provider, becoming burnout. The relation between homeless services and burnout has been critically underexplored, with no research carried out in Ireland to date, leaving a gap in the literature available. This lack of research added to the high influx of homeless people in Ireland provides the perfect scenario for our study. Through this research we would like to investigate the reason why providers of homeless services suffer burnout through the eyes of people that are suffering its effects.

RIGHT TO WITHDRAW?

You can decide to withdraw from the study at any point prior to the transcripts being anonymised without any consequence. You can contact the researcher to request this.

WHO SHOULD YOU CONTACT FOR FURTHER INFORMATION?

If you would like to receive any additional information or ask any other question related to this research, you can contact me on the following phone number 0868425508, or through email at x19126158@student.ncirl.ie

Looking forward to speaking with you,

Thank you very much for your time.

Marta Franchy

Appendix D - Consent Form

Consent to participate in the research

- I..... of my own accord agree to participate in this research.
- I understand that even if I agree to participate now, I can withdraw at any time or refuse to respond any question without any penalties.
- The purpose and nature of this research has been explained to me and the opportunity to ask questions has been given.
- I understand that my participation in this research requires to undertake an interview of 45 min to an hour.
- I consent to the recording of the interview
- I understand that all the information provided to this research will remain confidential
- I understand that my identity will remain anonymous, changing my name and concealing any identifiable information provided during the interview process that might reveal my identity.
- I understand that some extracts from my interview may be cited in the findings of the research carried out.
- I understand that any information provided to the researcher that could incur in the harm of myself or others must be reported to the pertinent authorities by the researcher.
- I understand that a copy of the interview recording, transcript and consent form will be kept in a password protected folder to which only the researcher has access.
- I understand that a copy of the interview recording, transcript and consent form will be stored until the result of the research have been published.
- I understand that all the information provided will be available for me to access it
- I understand that I can contact any of the parties involved in conducting this research for further clarification if needed

Marta Franchy Garcia (Researcher)

Tel. 0868425508, x19126158@student.ncirl.ie

Desmond Gibney (Supervisor)

desmond.gibney@ncirl.ie

Signature of research participant